

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Pesaran, Mohammad Hashem; Tosetti, Elisa

Working Paper Large panels with common factors and spatial correlations

CESifo Working Paper, No. 2103

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Pesaran, Mohammad Hashem; Tosetti, Elisa (2007) : Large panels with common factors and spatial correlations, CESifo Working Paper, No. 2103, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/26148

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

LARGE PANELS WITH COMMON FACTORS AND SPATIAL CORRELATIONS

M. HASHEM PESARAN ELISA TOSETTI

CESIFO WORKING PAPER NO. 2103 CATEGORY 10: EMPIRICAL AND THEORETICAL METHODS SEPTEMBER 2007

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com • from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

LARGE PANELS WITH COMMON FACTORS AND SPATIAL CORRELATIONS

Abstract

This paper considers the statistical analysis of large panel data sets where even after conditioning on common observed effects the cross section units might remain dependently distributed. This could arise when the cross section units are subject to unobserved common effects and/or if there are spill over effects due to spatial or other forms of local dependencies. The paper provides an overview of the literature on cross section dependence, introduces the concepts of time-specific weak and strong cross section dependence and shows that the commonly used spatial models are examples of weak cross section dependence. It is then established that the Common Correlated Effects (CCE) estimator of panel data model with a multifactor error structure, recently advanced by Pesaran (2006), continues to provide consistent estimates of the slope coefficient, even in the presence of spatial error processes. Small sample properties of the CCE estimator under various patterns of cross section dependence, including spatial forms, are investigated by Monte Carlo experiments. Results show that the CCE approach works well in the presence of weak and/or strong cross sectionally correlated errors. We also explore the role of certain characteristics of spatial processes in determining the performance of CCE estimators, such as the form and intensity of spatial dependence, and the sparseness of the spatial weight matrix.

JEL Code: C10, C31, C33.

Keywords: panels, Common Correlated Effects, strong and weak cross section dependence.

M. Hashem Pesaran Faculty of Economics University of Cambridge Sidgwick Avenue Cambridge, CB3 9DD United Kingdom mhp1@econ.cam.ac.uk

Elisa Tosetti Faculty of Economics University of Cambridge Sidgwick Avenue Cambridge, CB3 9DD United Kingdom

August 2007

We are grateful to Alexander Chudik and George Kapetanios for helpful comments and suggestions.

1 Introduction

Over the past few years there has been a growing literature, both empirical and theoretical, on econometric analysis of panel data models with cross sectionally dependent error processes. Cross correlations can be due to omitted common effects, spatial effects, or could arise as a result of interactions within socio-economic networks. Conditioning on variables specific to the cross section units alone does not deliver cross section error independence; an assumption required by the standard literature on panel data models. In the presence of such dependence, conventional panel estimators such as fixed or random effects can result in misleading inference and even inconsistent estimators. (Phillips and Sul, 2003). Further, conventional panel estimators will be inconsistent if regressors are correlated with the factors behind the error cross section dependence. (Pesaran, 2006). Correlation across units in panels has also serious drawbacks on commonly used panel unit root tests, since several of the existing tests assume independence (Levin, Lin and Chu, 2002; Im, Pesaran and Shin, 2003). As a result, when applied to cross sectional dependent panels, such unit root tests tend to have substantial size distortions (O'Connell, 1998). This potential problem has recently given major impetus to the research on panel unit root tests that allow for cross correlations (see, for example, Bai and Ng, 2004, Moon and Perron, 2004, and Pesaran, 2007). These and other related developments are reviewed in Breitung and Pesaran (2007).

In the case of panel data models where the cross section dimension, N, is small and the time series dimension, T, is large, the standard approach to cross section dependence is to consider the equations from different cross section units as a system of seemingly unrelated regression equations (SURE), and then estimate it by Generalized Least Squares techniques (Zellner, 1962). If the time series dimension is not sufficiently large, and in particular if N > T, the SURE approach is not feasible. The approach also fails to yield consistent estimators if the cross dependence is due to an observed common factor which is correlated with the included observed regressors.

Currently, there are two main strands in the literature for dealing with error cross section dependence in panels where N is large relative to T, namely the residual multifactor and the spatial econometric approaches. The multifactor approach assumes that the cross dependence can be characterized by a finite number of unobserved common factors, possibly due to economy-wide shocks that affect all units albeit with different intensities. Under this framework, the error term is a linear combination of few common time-specific effects with heterogeneous factor loadings plus an idiosyncratic (individual-specific) error term. Estimation of a panel with such multifactor residual structure can be addressed by full maximum likelihood procedure (Robertson and Symons, 2000), or by principal component analysis (Coakley, Fuertes and Smith, 2002). A major shortcoming of these techniques is that they are not applicable when the regressors are correlated with the common shocks, as in this case they lead to inconsistent estimators. Recently, Pesaran (2006) has suggested an estimation method, known as Common Correlated Effects (CCE), that consists of approximating the linear combinations of the unobserved factors by cross section averages of the dependent and explanatory variables and then running standard panel regressions augmented with the cross section averages. An advantage of this approach is that it yields consistent estimates also when the regressors are correlated with the factors. Monte Carlo studies have also shown that,

compared to other existing methods, CCE estimation is the most efficient and robust to alternative assumptions various hypothesis of non-stationarity of variables and factors (Coakley, Fuertes and Smith, 2006; Kapetanios and Pesaran, 2007).

The spatial approach assumes that the structure of cross section correlation is related to location and distance among units, defined according to a pre-specified metric. Proximity needs not be measured in terms of physical space, but can be defined using other types of metrics, such as economic (Conley, 1999, Pesaran, Schuermann and Weiner, 2004), policy, or social distance (Conley and Topa, 2002). Hence, cross section correlation is represented by means of a spatial process, which explicitly relates each unit to its neighbours (Whittle, 1954). Spatial econometric literature has suggested a number of processes for modeling spatial dependence. The most widely applied are the Spatial Moving Average (SMA) and the Spatial Error Component (SEC) specifications, and the Spatial Autoregressive (SAR) model. They differ in the range of dependence implied by their covariance matrices. Estimation of panels with spatially correlated errors is generally based on maximum likelihood techniques (Anselin, 1988), or on the generalized method of moments (Kelejian and Prucha, 1999; Conley, 1999). However, most of the work on spatial econometrics has so far been carried out in the context of a single cross section, or under strong cross section homogeneity and time invariance assumptions. While few authors consider heterogeneity and time dependency in spatial panels (Anselin, 1988; Kelejian et al., 2006), the study of local forms of correlation across units in more general framework of heterogeneous panel regressions still remains a challenging topic.

This paper considers estimation of a panel data model with common factors where the idiosyncratic errors could display spatial dependence. Specifically, we consider a panel data model where the error term is the sum of a multifactor structure and a spatial process. In this setting, we show that Pesaran's CCE approach continues to be valid, and under certain standard assumptions on the spatial error process provides consistent estimates of the slope coefficients. To this end, we introduce the concepts of time-specific weak and strong cross section dependence. We define a process to be cross sectionally weakly dependent at a given point in time if its weighted average at that time converges to it expectation in quadratic mean, as the cross section dimension is increased without bounds for all predetermined weights that satisfy certain 'granularity' conditions. If this condition does not hold, then the process is said to be cross sectionally strongly dependent. The distinctive feature of strong correlation is that it is pervasive, in the sense that it remains common to all units in a progressively larger cross sectional sample. Convergence properties of weighted averages is of great importance for the asymptotic theory of various estimators and tests commonly used in econometrics. From a more general perspective, the time-specific notions of weak and strong cross section dependence is also related to a consistent body of research in finance, the theory of asset pricing and portfolio choice (Chamberlain, 1983; Forni and Lippi, 2001).

We also investigate the nature of dependency conveyed by most widely applied processes for modeling contemporaneous correlation. We prove that common factor models provide a class of panel data models with both strong and weak dependence, while commonly used spatial processes are characterizations of weak cross section dependence.

We conclude the paper with a Monte Carlo study investigating the small sample performance of

the CCE estimator under alternative cross section error correlations. We examine the performance of the CCE estimator both when the errors display spatial dependence and when they are subject to unobserved common factors. The mixture case where both sources of cross section dependence exist is also considered. In particular, we show how certain characteristics of spatial processes, such as the spatial intensity and the sparseness of the spatial matrix, affect the performance of the CCE estimator.

The plan of the remainder of the paper is as follows. Section 2 sets out the panel regression model with common factors and spatial dependence and discusses the idea underlying the CCE approach. Section 3 introduces the concepts of strong and weak cross section dependence. Section 4 explore the relationship between the dependence structure of processes and the existence of dominant cross section units. Section 5 presents common factor models as examples of models with strong dependence. Section 6 investigates the nature of cross section dependence implied by most commonly used spatial processes. Section 7 introduces the CCE estimator in the context of panels with common factors and spatial dependence. Section 8 describes the Monte Carlo design and discusses the results. Finally, Section 9 provides some concluding remarks.

2 Panels with common factors and spatial dependence

Let y_{it} be the observation on the i^{th} cross section unit at time t for i = 1, 2, ..., N; t = 1, 2, ..., T, and suppose that it is generated as

$$y_{it} = \boldsymbol{\alpha}'_i \mathbf{d}_t + \boldsymbol{\beta}'_i \mathbf{x}_{it} + \boldsymbol{\gamma}'_i \mathbf{f}_t + e_{it}, \tag{1}$$

where \mathbf{d}_t is a $n \times 1$ vector of observed common effects, \mathbf{x}_{it} is a $k \times 1$ vector of observed individual specific regressors on the i^{th} cross section unit at time t, $\mathbf{f}_t = (f_{1t}, f_{2t}, ..., f_{mt})'$ is an *m*-dimensional vector of unobservable variables (known as common shocks/factors), $\boldsymbol{\gamma}_i = (\gamma_{1i}, \gamma_{2i}, ..., \gamma_{mi})'$ is the associated $m \times 1$ vector of factor loadings, and e_{it} is the unit-specific (idiosyncratic) error term. The common factors, \mathbf{f}_t , are viewed as shocks that simultaneously affect all cross section units, albeit with different degrees.

To model the correlation between the individual specific regressors, \mathbf{x}_{it} , and the common factors we suppose that

$$\mathbf{x}_{it} = \mathbf{A}'_i \mathbf{d}_t + \mathbf{\Gamma}'_i \mathbf{f}_t + \mathbf{v}_{it},\tag{2}$$

where \mathbf{A}_i and $\mathbf{\Gamma}_i$ are $n \times k$ and $m \times k$ factor loading matrices with fixed components, and \mathbf{v}_{it} is the individual component of \mathbf{x}_{it} , assumed to be distributed independently of the common factors $\mathbf{g}_t = (\mathbf{d}'_t, \mathbf{f}'_t)'$.

Equations (1) and (2) can be written more compactly as

$$\mathbf{z}_{it} = \begin{pmatrix} y_{it} \\ \mathbf{x}_{it} \end{pmatrix} = \mathbf{B}'_{i}\mathbf{d}_{t} + \mathbf{C}'_{i}\mathbf{f}_{t} + \mathbf{D}'_{i}\mathbf{u}_{it}, \qquad (3)$$

where

$$\begin{aligned} \mathbf{u}_{it} &= \begin{pmatrix} \mathbf{e}_{it} \\ \mathbf{v}_{it} \end{pmatrix}, \ \mathbf{D}_i = \begin{pmatrix} 1 & \mathbf{0} \\ \beta_i & \mathbf{I}_k \end{pmatrix}, \\ \mathbf{B}_i &= \begin{pmatrix} \alpha_i & \mathbf{A}_i \end{pmatrix} \mathbf{D}_i, \ \mathbf{C}_i = \begin{pmatrix} \gamma_i & \Gamma_i \end{pmatrix} \mathbf{D}_i. \end{aligned}$$

The idea underlying the CCE approach is that the unobserved factors can be well approximated by the cross section averages of \mathbf{z}_{it} . Hence, estimation can be carried out by least squares applied to auxiliary regressions where the observed regressors are augmented with these cross section averages plus the observed common factors, \mathbf{d}_t .

As a way of illustrating this result, consider a set of fixed weights $\{w_i\}$ that add up to unity and satisfy the condition $\sum_{i=1}^{N} w_i^2 = O(N^{-1})$. Let

$$\boldsymbol{\xi}_{it} = \mathbf{D}'_{i} \mathbf{u}_{it} = \begin{pmatrix} e_{it} + \boldsymbol{\beta}'_{i} \mathbf{v}_{it} \\ \mathbf{v}_{it} \end{pmatrix}.$$
(4)

Then the cross section average of (3), using the weights w_i yields

$$\bar{\mathbf{z}}_{wt} = \bar{\mathbf{B}}'_{w} \mathbf{d}_{t} + \bar{\mathbf{C}}'_{w} \mathbf{f}_{t} + \bar{\boldsymbol{\xi}}_{wt},\tag{5}$$

$$\bar{\mathbf{z}}_{wt} = \sum_{i=1}^{N} w_i \mathbf{z}_{it}, \ \bar{\boldsymbol{\xi}}_{wt} = \sum_{i=1}^{N} w_i \boldsymbol{\xi}_{it},$$
$$\bar{\mathbf{B}}_{w} = \sum_{i=1}^{N} w_i \mathbf{B}_i, \text{ and } \bar{\mathbf{C}}_{w} = \sum_{i=1}^{N} w_i \mathbf{C}_i.$$

If we assume

$$Rank(\bar{\mathbf{C}}_w) = m \le k+1, \text{ for all } N, \tag{6}$$

it follows that

$$\mathbf{f}_t = (ar{\mathbf{C}}_war{\mathbf{C}}_w')^{-1}ar{\mathbf{C}}_w\left[ar{\mathbf{z}}_{wt} - ar{\mathbf{B}}_w'\mathbf{d}_t - ar{m{\xi}}_{wt}
ight].$$

Therefore, \mathbf{f}_t can be approximated by linear combinations of $(\bar{\mathbf{z}}_{wt}, \mathbf{d}_t)$, if

$$\bar{\boldsymbol{\xi}}_{wt} \stackrel{q.m.}{\to} \boldsymbol{0}. \tag{7}$$

In such a case we obtain

$$\mathbf{f}_t - (\mathbf{C}\mathbf{C}')^{-1}\mathbf{C} \left[\bar{\mathbf{z}}_{wt} - \bar{\mathbf{B}}'_w \mathbf{d}_t \right] \stackrel{q.m.}{\to} 0, \text{ as } N \to \infty,$$
(8)

where

$$\bar{\mathbf{C}}_w \xrightarrow{p} \mathbf{C} = \tilde{\mathbf{\Gamma}} \begin{pmatrix} 1 & \mathbf{0} \\ \beta & \mathbf{I}_k \end{pmatrix}, \text{ as } N \to \infty,$$
(9)

 $\tilde{\mathbf{\Gamma}} = (E(\boldsymbol{\gamma}_i), E(\mathbf{\Gamma}_i))$, and $\boldsymbol{\beta} = \mathbf{E}(\boldsymbol{\beta}_i)$.¹ One of the main contributions of this paper is to investigate general conditions under which $\bar{\boldsymbol{\xi}}_{wt} \stackrel{q.m.}{\to} \mathbf{0}$, in particular in cases where the individual specific errors, e_{it} , display spatial dependence.

To this end we now introduce some definitions and notations to be used throughout the paper: $\lambda_1(\mathbf{A}) \geq ... \geq \lambda_n(\mathbf{A})$ are the eigenvalues of a matrix $\mathbf{A} \in \mathbb{M}_n$, where \mathbb{M}_n is the space of real $n \times n$ matrices. \mathbf{A}^- denotes a generalized inverse of \mathbf{A} . \mathbb{R} is the set of real numbers, \mathbb{N} the set of natural numbers, and \mathbb{Z} is the set of integers. The following definitions establish some useful terminologies (see, for example, Horn and Johnson (1985, p.35, pp.290-291).

Definition 1 Let $\mathbf{A} \in \mathbb{M}_n$. The spectral radius $\rho(\mathbf{A})$ of \mathbf{A} is

$$\rho(\mathbf{A}) \equiv \max_{1 \le i \le n} \left\{ \left| \lambda_i \left(\mathbf{A} \right) \right| \right\}.$$

Definition 2 Let $\mathbf{A} \in \mathbb{M}_n$. The column norm of \mathbf{A} is

$$\|\mathbf{A}\|_{c} = \max_{1 \le j \le n} \sum_{i=1}^{n} |a_{ij}|.$$

The row norm of \mathbf{A} is

$$\|\mathbf{A}\|_{r} = \max_{1 \le i \le n} \sum_{j=1}^{n} |a_{ij}|.$$

The Euclidean norm of \mathbf{A} is

$$\left\|\mathbf{A}\right\|_{2} = \left[Tr(\mathbf{A}\mathbf{A}')\right]^{1/2}.$$

Definition 3 Let $\{a_n\}$ be a non-stochastic sequence (Lee, 2002).

- (i) $\{a_n\}$ is bounded (denoted by O(1)) if there exist a finite constant K that does not depend on n such that $|a_n| \leq K$ for all n.
- (ii) $\{a_n\}$ is at most of order b_n (denoted by $O(b_n)$) if there exists a constant K independent of n such that $|a_n| \leq Kb_n$ for all n.
- (iii) $\{a_n\}$ is bounded away from zero at a rate b_n if there exist a positive sequence $\{b_n\}$ and a constant K > 0 independent of n such that $K \leq \frac{|a_n|}{b_n}$ for n sufficiently large.

We make the following assumptions on the common factors, their loadings, and the individual or unit specific errors:

ASSUMPTION 1: The $(n + m) \times 1$ vector $\mathbf{g}_t = (\mathbf{d}'_t, \mathbf{f}'_t)'$ is a covariance stationary process, with absolute summable autocovariances, distributed independently of $e_{it'}$ and $\mathbf{v}_{it'}$ for all i, t, t'.²

¹Pesaran also shows that the CCE continues to be applicable even if the rank condition, (6), is not satisfied. In this case the mathematical details are much more complicated.

 $^{^{2}}$ This assumption can be relaxed to allow for unit roots in the common factors, along the lines shown in Kapetanios, Pesaran and Yagamata (2006).

ASSUMPTION 2: The individual-specific errors e_{it} and $\mathbf{v}_{jt'}$ are distributed independently for all i, j, t and t', and for each i, \mathbf{v}_{it} follows a linear stationary process with absolute summable autocovariances given by

$$\mathbf{v}_{it} = \sum_{\ell=0}^\infty \Pi_{i\ell} oldsymbol{\zeta}_{i,t-\ell},$$

where for each i, ζ_{it} is a $k \times 1$ vector of serially uncorrelated random variables with mean zero, the variance matrix \mathbf{I}_k , and finite fourth-order cumulants. The coefficient matrices $\Pi_{i\ell}$ satisfy the condition

$$Var(\mathbf{v}_{it}) = \sum_{\ell=0}^{\infty} \mathbf{\Pi}_{i\ell} \mathbf{\Pi}'_{i\ell} = \mathbf{\Sigma}_{vi} \leq \bar{\mathbf{\Sigma}}_{v} < \infty,$$

for all i and some constant matrix $\bar{\Sigma}_{v}$, where Σ_{vi} is a positive definite matrix.

ASSUMPTION 3: The unobserved factor loadings, γ_i and Γ_i are bounded, i.e. $\|\gamma_i\|_2 < K$ and $\|\Gamma_i\|_2 < K$, for all *i*.

Assumption 4: The slope coefficients β_i follow the random coefficient model

$$\boldsymbol{\beta}_i = \boldsymbol{\beta} + \boldsymbol{v}_i, \ \boldsymbol{v}_i \sim IID(0, \boldsymbol{\Omega}_v) \text{ for } i = 1, ..., N_i$$

where $\|\boldsymbol{\beta}\| < K$, $\|\boldsymbol{\Omega}_{\boldsymbol{v}}\|_2 < K$, $\boldsymbol{\Omega}_{\boldsymbol{v}}$ is a symmetric non negative definite matrix, and the random deviations, \boldsymbol{v}_i , are distributed independently of $\boldsymbol{\gamma}_j$, $\boldsymbol{\Gamma}_j$, e_{jt} , \mathbf{v}_{jt} , and \mathbf{g}_t for all i, j and t.

Finally, we assume that the individual specific errors e_{it} follow a *Spatial Autoregressive* (SAR) processes (Cliff and Ord, 1973, 1981):

$$e_{it} = \delta \bar{e}_{it} + \varepsilon_{it},\tag{10}$$

where δ is a scalar parameter, and \bar{e}_{it} is the spatial lag of e_{it} , defined by

$$\bar{e}_{it} = \sum_{j=1}^{N} s_{ij,t} e_{jt},\tag{11}$$

and ε_{it} are cross sectionally independently distributed random variables with mean zero and finite variances, σ_i^2 . But for each *i* the errors ε_{it} can be serially correlated.

Similar spatial models can also be specified for the individual-specific errors of the \mathbf{x}_{it} equation. But to keep the exposition simple we shall not be concerned with this problem explicitly, and assume that \mathbf{v}_{it} is cross sectionally weakly dependent (see below). As we shall see this assumption does allow for the possibility of spatial dependence across the individual specific errors of the \mathbf{x}_{it} equation.

In (11), $s_{ij,t}$ is the generic element of a $N \times N$ matrix \mathbf{S}_t , known as spatial weight matrix, which provides information on the ordering of cross sectional units and the various network or neighborhood linkages at time t (Anselin, 1988, 2002). In matrix form, the spatial model (10) can be written as

$$\mathbf{e}_t = \delta \mathbf{S}_t \mathbf{e}_t + \boldsymbol{\varepsilon}_t, \tag{12}$$

where $\mathbf{e}_t = (e_{1t}, ..., e_{Nt})'$ and $\boldsymbol{\varepsilon}_t = (\varepsilon_{1t}, ..., \varepsilon_{Nt})'$. Hence, the error term associated to a cross section unit is modelled as a weighted average of errors corresponding to its neighbouring cross section units, plus a white noise.

In the following, we introduce the notions of *time-specific* weak and strong cross section dependence for a stochastic process, on the basis of the convergence properties of its cross section average, as the number of units in the average is allowed to increase without bound. We investigate the conditions under which the CCE estimator continues to be valid in the presence of spatial dependence such as the SAR process defined by (12).

3 Weak and strong cross section dependence

In this section, we study the structure of correlation of a double index process $\{z_{it}, i \in S, t \in T\}$ where $S \subseteq \mathbb{N}, T \subseteq \mathbb{Z}$, and z_{it} are random variables defined on a probability space (Ω, \mathcal{F}, P) . In this paper the index t refers to an ordered set, the time, while the index i indicates the units of an unordered population. Our primary focus is on characterizing the correlation pattern of the double index process $\{z_{it}\}$ over the cross sectional dimension. To this end, we make the following assumptions:

ASSUMPTION 5: Let $\mathbf{w}_t = (w_{1t}, ..., w_{Nt})'$ be a vector of pre-determined weights and assume that the following 'granularity' conditions hold for all $t \in \mathcal{T}$:

$$\|\mathbf{w}_t\|_2 = O\left(N^{-\frac{1}{2}}\right) \tag{13}$$

$$\frac{w_{jt}}{\|\mathbf{w}_t\|_2} = O\left(N^{-\frac{1}{2}}\right) \quad \text{for any } j \le N \tag{14}$$

Also, although not necessary for our analysis, it is convenient to assume that the following normalization condition holds

$$\sum_{j=1}^{N} w_{jt} = 1.$$
(15)

ASSUMPTION 6: Let \mathcal{I}_t be the information set at time t containing at least $\mathbf{z}_t, \mathbf{z}_{t-1}, ...$ and $\mathbf{w}_t, \mathbf{w}_{t-1}, ...,$ where $\mathbf{z}_t = (z_{1t}, ..., z_{Nt})'$ and $\mathbf{w}_t = (w_{1t}, ..., w_{Nt})'$. For each $t \in \mathcal{T}$, \mathbf{z}_t has conditional mean and variance

$$E\left(\mathbf{z}_{t} | \mathcal{I}_{t-1}\right) = \mathbf{0}, \tag{16}$$

$$Var\left(\mathbf{z}_{t} | \mathcal{I}_{t-1}\right) = \boldsymbol{\Sigma}_{t}, \qquad (17)$$

where Σ_t is a $N \times N$ symmetric, nonnegative definite matrix, with generic $(i, j)^{th}$ element $\sigma_{ij,t}$, and such that $0 < \sigma_{ii,t} \leq K$, for i = 1, ..., N, where K is a finite constant independent of N.

Assumption 5, known in finance as the granularity condition, ensures that the weights $\{w_{it}\}$ is not dominated by a few of the cross section units. In Assumption 6 we impose some regularity conditions on the time series properties of $\{z_{it}\}$. Assumption 6 is also standard in finance and specifies that \mathbf{z}_t has conditional means and variances. The first part, (16), can be relaxed to

 $E(\mathbf{z}_t | \mathcal{I}_{t-1}) = \boldsymbol{\mu}_{t-1}$, with $\boldsymbol{\mu}_{t-1}$ being a pre-determined function of the elements of \mathcal{I}_{t-1} . But to keep the exposition simple and without loss of generality we have set $\boldsymbol{\mu}_{t-1} = \mathbf{0}$.

Consider now the weighted averages, $\bar{z}_{wt} = \sum_{i=1}^{N} w_{i,t-1} z_{it} = \mathbf{w}'_{t-1} \mathbf{z}_t$, for $t \in \mathcal{T}$, where \mathbf{z}_t and \mathbf{w}_{t-1} satisfy Assumptions 5 and 6. We are interested in the limiting behaviour of $\mathbf{w}'_{t-1} \mathbf{z}_t$ for each $t \in \mathcal{T}$ as $N \to \infty$.

Definition 4 (weak dependence) The process $\{z_{it}\}$ is said to be cross sectionally weakly dependent (CWD) at a given point in time $t \in \mathcal{T}$, if for all weight vectors, \mathbf{w}_{t-1} , satisfying the granularity conditions (13)-(14) we have

$$\lim_{N \to \infty} Var(\mathbf{w}_{t-1}' \mathbf{z}_t | \mathcal{I}_{t-1}) = 0.$$
(18)

If (18) holds for all $t \in \mathcal{T}$, then we say that the process is cross sectionally weakly dependent.

Proposition 5 The process $\{z_{it}\}$ is CWD at a point in time $t \in \mathcal{T}$ if $\lambda_1(\Sigma_t)$ is bounded.

Proof. We note that by the Rayleigh-Ritz theorem³

$$Var(\mathbf{w}_{t-1}'\mathbf{z}_t | \mathcal{I}_{t-1}) = \mathbf{w}_{t-1}' \mathbf{\Sigma}_t \mathbf{w}_{t-1} \le \left(\mathbf{w}_{t-1}' \mathbf{w}_{t-1}\right) \lambda_1\left(\mathbf{\Sigma}_t\right).$$
(19)

Since $\lambda_1(\Sigma_t)$ is bounded for all N, then under the granularity conditions (13)-(14) we have

$$\lim_{N\to\infty} Var(\mathbf{w}_{t-1}'\mathbf{z}_t | \mathcal{I}_{t-1}) = 0$$

A particular form of a CWD process arises when pairwise correlations take non-zero values only across finite subsets of units that do not spread widely as sample size increases. A similar case occurs in spatial processes, where for example local dependency exists only among adjacent observations. In Section 6 we will characterize the cross correlations implied by most commonly used spatial processes.

A number of remarks concerning the above concept of CWD is in order.

Remark 6 The notion of weak dependence does not necessarily involve an ordering of the observations or the specification of a distance metric.

Remark 7 It is clear from (19) that if $\lambda_1(\Sigma_t) = O(N^{1-\epsilon})$ for any $\epsilon \in (0,1)$, then

$$\lim_{N \to \infty} \left(\mathbf{w}_{t-1}' \mathbf{w}_{t-1} \right) \lambda_1 \left(\mathbf{\Sigma}_t \right) = 0,$$

and the underlying process will be CWD. Hence, the bounded eigenvalue condition discussed in the literature is sufficient but not necessary for CWD. According to our definition a process could be CWD even if its maximum eigenvalue is rising with N, so long as its rate of increase is bounded appropriately.

³See Horn and Johnson (1985, p.176).

Remark 8 The concept of "idiosyncratic" developed in Forni and Lippi (2001) can be derived from our notion of CWD as a special case. Forni and Lippi (2001) define as "idiosyncratic" any double sequence, $\{z_{it}\}$, if its weighted average computed both over cross section units and across time converges to zero in quadratic mean, for all sets of weights satisfying the granularity condition (13). They show that the boundedness of the largest eigenvalue of the spectral density matrix of $\{z_{it}\}$ (at all frequencies) is necessary and sufficient for the process to be idiosyncratic. To ensure the existence of the spectral density, Forni and Lippi assume that the underlying time series processes are stationary with absolutely summable autocovariances. In contrast, in our development of the CWD, we consider the asymptotic behaviour of the weighted averages at each point in time, which does not require any regularity assumptions on the time series properties of the underlying processes beyond those set out in Assumption 6.

Remark 9 Also in establishing the equivalence of the bounded eigenvalue conditions and the idiosyncratic property, Forni and Lippi consider weights that are not pre-determined, and in principle could depend on the elements of Σ (taken by Forni and Lippi to be time invariant). By focusing on pre-determined weights, as we shall see below, we are able to provide a more natural link to the literatures on panel estimation and inference and the portfolio decision theory.

If condition (18) is not satisfied, then the incidence of non-zero pair-wise correlations across the elements of Σ_t would be pervasive. More specifically, we say the process $\{z_{it}\}$ is cross sectionally strongly dependent (CSD) at a point in time $t \in T$, if there exists a vector of weights, \mathbf{w}_{t-1} , that satisfy (13)-(14), and a constant K independent of N such that for N sufficiently large

$$Var(\mathbf{w}_{t-1}'\mathbf{z}_t | \mathcal{I}_{t-1}) \ge K > 0.$$

$$(20)$$

Proposition 10 Let $\{z_{it}\}$ be cross sectionally strongly dependent at a point in time $t \in \mathcal{T}$. Then $\lambda_1(\Sigma_t)$ and $\|\Sigma_t\|_c$ are bounded away from zero at rate $(\mathbf{w}'_{t-1}\mathbf{w}_{t-1})^{-1}$.

Proof. For a strongly dependent process, under the granularity conditions (13)-(14) we can find a set of weights \mathbf{w}_{t-1} such that

$$0 < K \le Var(\mathbf{w}_{t-1}'\mathbf{z}_t | \mathcal{I}_{t-1}) \le \left(\mathbf{w}_{t-1}'\mathbf{w}_{t-1}\right) \lambda_1\left(\mathbf{\Sigma}_t\right),\tag{21}$$

which proves that $\lambda_1(\mathbf{\Sigma}_t)$ diverges to infinity at least at rate $(\mathbf{w}'_{t-1}\mathbf{w}_{t-1})^{-1}$. Further, we know that⁴

$$\lambda_1\left(\mathbf{\Sigma}_t\right) \leq \left\|\mathbf{\Sigma}_t\right\|_c$$
 .

It follows from (21) that

$$0 < K \le \left(\mathbf{w}_{t-1}'\mathbf{w}_{t-1}\right) \lambda_1\left(\mathbf{\Sigma}_t\right) \le \left(\mathbf{w}_{t-1}'\mathbf{w}_{t-1}\right) \left\|\mathbf{\Sigma}_t\right\|_c,$$

which proves that $\|\mathbf{\Sigma}_t\|_c$ must tend to infinity at least at rate $(\mathbf{w}'_{t-1}\mathbf{w}_{t-1})^{-1}$.

⁴See Horn and Johnstone (1985, pp. 297-298).

From the above proposition, both the spectral radius and the column norm of the covariance matrix of a CSD process are unbounded in N. This result for a CSD process is similar to the condition of not absolutely summable autocorrelations that characterizes time series processes with strong autocorrelation (Robinson, 2003). We observe that for any process $\{z_{it}\}$ either weak or strongly correlated, $\lambda_1(\Sigma_t)$ and $\|\Sigma_t\|_c$ cannot diverge to infinity at a rate larger than N. Indeed, since

$$\lambda_1\left(\boldsymbol{\Sigma}_t\right) \leq \sum_{i=1}^N \sigma_{ii,t}, \text{ and } \|\boldsymbol{\Sigma}_t\|_c \leq \left(\max_{1 \leq j \leq N} \sigma_{jj,t}^{1/2}\right) \sum_{i=1}^N \sigma_{ii,t}^{1/2},$$

and given that by Assumption 6 $\sigma_{ii,t}$, the diagonal elements of Σ_t , are finite, it follows that $\lambda_1(\Sigma_t)$ and $\|\Sigma_t\|_c$ approach infinity at most at rate N.

Remark 11 The above concepts of weak and strong cross section dependence are also related to the notion of diversifiability provided by the asset pricing theory (Chamberlain, 1983). In this context, Σ_t represents the covariance matrix of a vector of random returns on N different assets, and $w_{i,t-1}$, for i = 1, 2, ..., N, denote the proportion of investor's wealth allocated to the i^{th} asset. From Definition 4 it follows that the part of asset returns that is weakly dependent will be fully diversified by portfolios constructed using \mathbf{w}_{t-1} as the portfolio weights, and as $N \to \infty$. The part of asset returns that is strongly dependent can only be diversified with portfolio weights that are dependent on the factor loadings. Indeed, assume for simplicity that returns $\{r_{it}\}$ have the one-factor structure

$$r_{it} = \gamma_i f_t + e_{it},$$

where $f_t \sim IID(0,1)$, and $\{e_{it}\}$ is a CWD process distributed independently of f_t and γ_i . Consider the portfolio weights

$$w_i^* = \frac{1}{N} \left[1 - \frac{\mu_{\gamma}}{s_{\gamma}^2} \left(\gamma_i - \bar{\gamma} \right) \right],$$

where

$$\mu_{\gamma} = \frac{1}{N} \sum_{i=1}^{N} \gamma_i, \ s_{\gamma}^2 = N^{-1} \sum_{i=1}^{N} \left(\gamma_i - \mu_{\gamma} \right)^2 > 0.$$

It is then easily verified that these weights satisfy the granularity conditions, (13)-(14), and since $\{e_{it}\}$ is CWD then

$$\lim_{N \to \infty} Var\left(\Sigma_{i=1}^{N} w_i^* r_{it} \left| f_t \right.\right) = f_t^2 \lim_{N \to \infty} \left(\Sigma_{i=1}^{N} w_i^* \gamma_i\right)^2.$$

However

$$\Sigma_{i=1}^{N} w_i^* \gamma_i = 0, \text{ for all } N,$$

and the variance of the portfolio return, $\rho_{w^*t} = \sum_{i=1}^N w_i^* r_{it}$, tends to zero as $N \to \infty$. Note that for all $\mu_{\gamma} \neq 0$, the underlying return $\{r_{it}\}$ is a CSD process.

We conclude this section with two results concerning the relationship between strongly correlated and weakly correlated variables. Following Definition 4, we say that two processes $\{z_{it,a}\}$ and $\{z_{it,b}\}$ are weakly correlated if $\lim_{N\to\infty} E(\bar{z}_{wt,a}\bar{z}_{wt,b} | \mathcal{I}_{t-1}) = 0$, for all sets of weights at time t that satisfy the granularity conditions. The next proposition considers correlation of two processes with different cross dependence structures. We then investigate the correlation structure of linear combinations of strongly correlated and weakly correlated variables.

Proposition 12 Suppose that $\{z_{it,a}\}$ and $\{z_{it,b}\}$ are CSD and CWD processes, respectively. Then for all sets of weights $\{w_{i,t-1}^a\}$ and $\{w_{i,t-1}^b\}$ satisfying conditions (13)-(14) we have

$$\lim_{N \to \infty} E(\bar{z}_{wt,a} \bar{z}_{wt,b} | \mathcal{I}_{t-1}) = 0.$$

Proof. Let $\{w_{i,t-1}^a\}$ and $\{w_{i,t-1}^b\}$ be two sets of weights satisfying conditions (13)-(14). For $t \in \mathcal{T}$, we have

$$\left[E(\bar{z}_{wt,a}\bar{z}_{wt,b} | \mathcal{I}_{t-1})\right]^2 \le E(\bar{z}_{wt,a}^2 | \mathcal{I}_{t-1})E(\bar{z}_{wt,b}^2 | \mathcal{I}_{t-1}).$$

Since $z_{it,a}$ is a CSD process then

$$\lim_{N \to \infty} E(\bar{z}_{wt,a}^2 | \mathcal{I}_{t-1}) > K > 0,$$

where K is a finite constant. Also from (18), and considering that $z_{it,b}$ is a CWD process we have

$$\lim_{N \to \infty} E(\bar{z}_{wt,b}^2 \,| \mathcal{I}_{t-1}) = 0.$$

Therefore, for all sets of weights satisfying (13)-(14), we obtain

$$\lim_{N \to \infty} E(\bar{z}_{wt,a} \bar{z}_{wt,b} | \mathcal{I}_{t-1}) = 0.$$

Proposition 13 Consider two independent processes $\{z_{it,a}\}$ and $\{z_{it,b}\}$, and their linear combinations defined by

$$z_{it,c} = \beta_a z_{it,a} + \beta_b z_{it,b},\tag{22}$$

where β_a and β_b are non-zero fixed coefficients. Then the following statements hold:

- (i) Suppose $\{z_{it,a}\}$ and $\{z_{it,b}\}$ are CSD, then $\{z_{it,c}\}$ is CSD,
- (ii) Suppose $\{z_{it,a}\}$ and $\{z_{it,b}\}$ are CWD, then $\{z_{it,c}\}$ is CWD,
- (iii) Suppose $\{z_{it,a}\}$ is CSD and $\{z_{it,b}\}$ is CWD, then $\{z_{it,c}\}$ is CSD.

Proof. Let $\Sigma_{t,a}$ and $\Sigma_{t,b}$ be the covariance matrices of $z_{t,a} = (z_{1t,a}, ..., z_{Nt,a})'$ and $z_{t,b} = (z_{1t,b}, ..., z_{Nt,b})'$, and $\Sigma_{t,c}$ the covariance of their linear combination that is, given the assumption of independence between $z_{t,a}$ and $z_{t,b}$

$$\boldsymbol{\Sigma}_{t,c} = \beta_a^2 \boldsymbol{\Sigma}_{t,a} + \beta_b^2 \boldsymbol{\Sigma}_{t,b}$$

The variance of the weighted average $\mathbf{w}'_{t-1}\mathbf{z}_{t,c}$ satisfies

$$Var(\mathbf{w}_{t-1}'\mathbf{z}_{t,c} | \mathcal{I}_{t-1}) \ge \beta_j^2 Var(\mathbf{w}_{t-1}'\mathbf{z}_{t,j} | \mathcal{I}_{t-1}), \quad j = a, b,$$

which implies that, if there exists a weights vector \mathbf{w}_{t-1} satisfying the granularity conditions such that either $Var(\mathbf{w}'_{t-1}\mathbf{z}_{t,a}|\mathcal{I}_{t-1})$ or $Var(\mathbf{w}'_{t-1}\mathbf{z}_{t,b}|\mathcal{I}_{t-1})$ or both are nonzero, then also $Var(\mathbf{w}'_{t-1}\mathbf{z}_{t,c}|\mathcal{I}_{t-1})$ is bounded away from zero and $\{z_{it,c}\}$ is cross sectionally strongly dependent (this proves (i) and (iii)). Also, by Weyl's theorem⁵, the largest eigenvalue of $\Sigma_{t,c}$ satisfies

$$\lambda_{1}\left(\boldsymbol{\Sigma}_{t,c}\right) \leq \beta_{a}^{2}\lambda_{1}\left(\boldsymbol{\Sigma}_{t,a}\right) + \beta_{b}^{2}\lambda_{1}\left(\boldsymbol{\Sigma}_{t,b}\right)$$

which establishes that if both $\lambda_1(\Sigma_{t,a})$ and $\lambda_1(\Sigma_{t,b})$ are bounded, then also $\lambda_1(\Sigma_{t,c})$ is bounded in N, and hence $\{z_{it,c}\}$ is cross sectionally weakly correlated (this proves (ii)).

The above result can be generalized to linear functions of more than two processes. In general, linear combinations of independent processes that are strongly (weakly) correlated is strongly (weakly) dependent, while linear combinations of a finite number of weakly and strongly correlated processes is strongly correlated, since on aggregation only terms involving the strong component will be of any relevance. This result will be employed in Section 5.

4 Strong dependence and dominant effects

In this section we introduce the notion of dominant effects in the covariance matrix of strongly dependent processes. We then explore the relationship between dominant units and the asymptotic behaviour of eigenvalues of the underlying covariance matrix. The discussion will introduce us to multifactor models which will be broadly investigated in Section 5.

Definition 14 Let $\mathbf{A} = (a_{ij}) \in \mathbb{M}_n$ with finite elements, a_{ij} . We say that \mathbf{A} has m dominant columns if there exists a m-dimensional subset $\mathcal{J}(m)$ of the index set $\mathcal{J} = \{1, 2, ..., n\}$ such that

(i)
$$\sum_{i=1}^{N} |a_{ij}| = O(N)$$
, for $j \in \mathcal{J}(m)$

(ii)
$$\sum_{i=1}^{N} |a_{ij}| = O(1)$$
 for $j \in \mathcal{J} \cap \mathcal{J}(m)^c$.

It also follows that matrix $\mathbf{A} \in \mathbb{M}_n$ will have *m* dominant rows if \mathbf{A}' has *m* dominant columns according to the above definition.

Clearly, if the conditional variance Σ_t of a process $\{z_{it}\}$ has *m* dominant columns, it will also have *m* dominant rows. In this case, the units corresponding to the unbounded columns and rows have a strong relationship with all other units. This situation arises in a number of empirical applications, in particular when it is possible to identify few dominant individuals (regions), the leaders, that influence many others, the followers. For example, in the context of global macroeconomic modelling, few large economies, such as US, strongly affect small economies, yet impact of small countries on the rest of the world as a whole is negligible (Pesaran and Chudik, 2007). In the study of price competition among firms, it is possible to consider not only localized forms of rivalry, where firms compete only with their close neighbours, but also a global or monopolistic competition, in which all products compete with one or few products of a particular brand (Pinkse et al., 2002). A

⁵See Horn and Johnson (1985, p. 181).

further example can be taken from the housing market in a country such as England, where house price variations in few major urban centres like London have an impact on the dynamics of prices in all other smaller and medium-sized cities in the country.

In Proposition 10 we have proved that under CSD the largest eigenvalue of the covariance matrix diverges to infinity at the rate of N. Further, we have seen that this implies that Σ_t has (at least) one dominant cross section unit. Under some general conditions the reverse is also true, that is if Σ_t has unbounded column norm, then the largest eigenvalue of Σ_t is also unbounded in N. Since Σ_t is a symmetric positive semi-definite matrix it can be decomposed as $\Sigma_t = \mathbf{R}_t \mathbf{R}'_t$, where \mathbf{R}_t is an $N \times r$ ($r \leq N$) matrix with bounded row norm. The following inequalities hold⁶

$$\lambda_1(\mathbf{\Sigma}_t) \leq \|\mathbf{\Sigma}_t\|_c \leq \|\mathbf{R}_t\|_c \|\mathbf{R}_t\|_r, \qquad (23)$$

$$\frac{\|\mathbf{R}_t\|_c}{\sqrt{N}} \leq [\lambda_1(\boldsymbol{\Sigma}_t)]^{1/2}.$$
(24)

From (24), if $\|\mathbf{R}_t\|_c$ is bounded away from zero at a rate greater than \sqrt{N} , then $\lambda_1(\mathbf{\Sigma}_t)$ also diverges to infinity. Note that if the largest eigenvalue is bounded, the column norm of \mathbf{R}_t (and hence of $\mathbf{\Sigma}_t$) can still diverge to infinity at a rate at most equal to \sqrt{N} . As we will see in Section 6 below, this result has a nice intuitive interpretation when dealing with spatial processes, where the \mathbf{R}_t matrix is a function of the spatial weight matrix.

We now consider the case where a strongly dependent process has more than one characteristic root that increases with N. As we will see in the following section, this is the case of common factor models (Chamberlain and Rothschild, 1983, Forni and Lippi, 2001). The following theorem holds.

Theorem 15 Consider a process $\{z_{it}\}$ which is cross sectionally strongly dependent at time $t \in \mathcal{T}$. Let m be a fixed number. Then the following statements hold:

- (i) Suppose $\lambda_m(\Sigma_t) = O(N)$, then the process underlying Σ_t has at least m dominant cross section units
- (ii) Suppose Σ_t has m dominant cross section units, then $\lambda_j(\Sigma_t) = O(N)$, for $1 \le j \le m$, and $\lambda_{m+1}(\Sigma_t)$ is bounded.

Proof. Suppose for simplicity that the columns and rows of Σ_t are arranged so that first column vector has the largest absolute sum norm, the second column vector has the second largest absolute sum norm, and so on. Call Σ_t^1 the matrix obtained from Σ_t by deleting its first row and first column. Let Σ_t^2 be the the matrix obtained from Σ_t^1 by deleting its first row and first column, Σ_t^3 that obtained dropping the first row and column of Σ_t^2 and so forth, to obtain the set of interlacing matrices $\Sigma_t^1, ..., \Sigma_t^m$. From the interlacing eigenvalues theorem (see Horn and Johnson, 1985; p. 186) we know that

$$\lambda_m(\boldsymbol{\Sigma}_t^1) \leq \lambda_m(\boldsymbol{\Sigma}_t) \leq \lambda_{m-1}(\boldsymbol{\Sigma}_t^1) \leq \dots \leq \lambda_2(\boldsymbol{\Sigma}_t) \leq \lambda_1(\boldsymbol{\Sigma}_t^1) \leq \lambda_1(\boldsymbol{\Sigma}_t)$$

⁶For (23) see Bernstein (2005, p.351), and for (24) see Bernstein (2005, p.368, eq. xiv).

From the above relation, the following inequalities can be established

$$\lambda_m(\mathbf{\Sigma}_t) \le \lambda_{m-1}(\mathbf{\Sigma}_t^1) \le \lambda_{m-2}(\mathbf{\Sigma}_t^2) \le \dots \le \lambda_2(\mathbf{\Sigma}_t^{m-2}) \le \lambda_1(\mathbf{\Sigma}_t^{m-1})$$
(25)

$$\lambda_{m+1}(\boldsymbol{\Sigma}_t) \le \lambda_m(\boldsymbol{\Sigma}_t^1) \le \lambda_{m-1}(\boldsymbol{\Sigma}_t^2) \le \dots \le \lambda_2(\boldsymbol{\Sigma}_t^{m-1}) \le \lambda_1(\boldsymbol{\Sigma}_t^m)$$
(26)

$$\lambda_1(\mathbf{\Sigma}_t^{m-1}) \le \dots \le \lambda_1(\mathbf{\Sigma}_t^2) \le \lambda_1(\mathbf{\Sigma}_t^1) \le \lambda_1(\mathbf{\Sigma}_t)$$
(27)

Also, recall that

$$\lambda_1(\mathbf{\Sigma}_t^j) \le \left\|\mathbf{\Sigma}_t^j\right\|_c \tag{28}$$

Suppose first that $\lambda_m(\Sigma_t) = O(N)$. Then from (25) $\lambda_1(\Sigma_t^{m-1})$ is also unbounded, implying (from (28)) that $\|\Sigma_t^{m-1}\|_c = O(N)$. Hence, Σ_t has at least m dominant units, which proves (i). Vice versa, suppose that Σ_t has m dominant units. Then we know from (24) that Σ_t has at least one eigenvalue unbounded in N. Further, note that, by the definition of dominant units, $\|\Sigma_t^m\|_c$ is bounded, and hence, from (28), $\lambda_1(\Sigma_t^m)$ is also bounded. From (26) it follows that $\lambda_{m+1}(\Sigma_t)$ is bounded, which proves (ii).

Note that in several cases the number of column vectors in Σ_t having unbounded sums largely exceeds the number of unbounded characteristic roots. In the extreme case, Σ_t could have N dominant units, with only one eigenvalues exploding to infinity, as in the following example of equicorrelation

$$\Sigma_{t} = \sigma^{2} \begin{pmatrix} 1 & \theta & \dots & \theta & \theta \\ \theta & 1 & \dots & \theta & \theta \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \theta & \theta & \dots & 1 & \theta \\ \theta & \theta & \dots & \theta & 1 \end{pmatrix},$$
(29)

where $|\theta| < 1$. In this case all column sums are unbounded. However, the characteristic roots of the above matrix are

$$\lambda_1(\boldsymbol{\Sigma}_t) = \sigma^2 [1 + (N-1)\theta]$$

$$\lambda_j(\boldsymbol{\Sigma}_t) = \sigma^2 (1-\theta) \text{ for } j = 2, ..., N,$$

namely only the largest eigenvalue is unbounded in N. In the next section we will see that processes with a covariance matrix like (29) can be well represented by the means of common factor models.

5 Common factor models

Originally proposed in the psychometric literature (Spearman, 1904), factor models are extensively used in macroeconomics and finance to represent the evolution of large cross sectional samples with strong co-movements. Panels with common factors have been applied to characterize the dynamic of stock and bond returns (Chamberlain and Rothschild 1983; Connor and Korajczyk, 1993; Kapetanios and Pesaran, 2007), and in macroeconomics to summarize the empirical content of a large number of variables by a small set of factors (Stone, 1947; Forni and Reichlin, 1998; Gregory and Head, 1999; Stock and Watson, 2002).

Factor models are typically used in the literature to represent CSD processes. However, according to our definition there are special cases where the factor representation could yield a CWD process. To see this, consider the following multifactor process for $\{z_{it}\}$

$$z_{it} = u_{it} + e_{it}, \quad i = 1, \dots, N, \tag{30}$$

$$u_{it} = \gamma_{i1} f_{1t} + \gamma_{i2} f_{2t} + \dots + \gamma_{im} f_{mt}, \qquad (31)$$

where the number of factors, m, is assumed to be smaller than the number of cross sectional units, N. The factors, $f_{\ell t}$, and their loadings, $\gamma_{i\ell}$, satisfy Assumptions 1 and 3, respectively. Without loss of generality we assume that $E(f_{\ell t}^2 | \mathcal{I}_{t-1}) = 1$ and $E(f_{\ell t} f_{pt} | \mathcal{I}_{t-1}) = 0$, for $\ell \neq p = 1, 2, ..., m$. We also make the following standard assumption to distinguish the common factors from the individual-specific component, $\mathbf{e}_t = (e_{1t}, e_{2t}, \dots, e_{Nt})'$, with $Var(\mathbf{e}_t | \mathcal{I}_{t-1}) = \boldsymbol{\Sigma}_{et}$.

ASSUMPTION 7: For a given $t \in \mathcal{T}$, $\lambda_1(\Sigma_{et})$ is bounded, namely $\{e_{it}\}$ is a CWD process.

Theorem 16 Consider the factor model (30)-(31), and suppose that assumption 1, 3 and 7 hold. The process $\{z_{it}\}$ is cross sectionally weakly dependent at a given point in time $t \in \mathcal{T}$, if and only if $\lim_{N\to\infty} \bar{\gamma}_{w_{t-1},\ell} = 0$, for $\ell = 1, ..., m$, and for all weights $\{w_{i,t-1}\}$ satisfying (13)-(14).

Proof. From Definition 1, z_{it} is CWD for at a given point in time $t \in \mathcal{T}$, if

$$\lim_{N \to \infty} Var(\bar{z}_{wt} | \mathcal{I}_{t-1}) = \lim_{N \to \infty} \left[Var(\bar{u}_{wt} | \mathcal{I}_{t-1}) + Var(\bar{e}_{wt} | \mathcal{I}_{t-1}) \right] = 0.$$

Since e_{it} is CWD it immediately follows that $\lim_{N\to\infty} Var(\bar{e}_{wt}|\mathcal{I}_{t-1}) = 0$. Consider now the first component of z_{it} and note that under our assumptions

$$\lim_{N \to \infty} Var(\bar{u}_{wt} | \mathcal{I}_{t-1}) = \lim_{N \to \infty} \sum_{i=1}^{N} \sum_{j=1}^{N} w_{i,t-1} w_{j,t-1} \left(\gamma_{i1} \gamma_{j1} + ... + \gamma_{im} \gamma_{jm} \right)$$
$$= \lim_{N \to \infty} \left(\bar{\gamma}_{w_{t-1},1}^2 + ... + \bar{\gamma}_{w_{t-1},m}^2 \right)$$

where $\bar{\gamma}_{w_{t-1},\ell} = \sum_{i=1}^{N} w_{i,t-1} \gamma_{i\ell}$. This establishes that weak cross section dependence of z_{it} is equivalent to $\lim_{N\to\infty} \bar{\gamma}_{w_{t-1},\ell} = 0$, for $\ell = 1, 2, ..., m$, for all sets of weights satisfying (13)-(14). As a consequence, if z_{it} is CWD then weighted averages of the factor loadings converge to zero as

 $N \to \infty$. Conversely, for a strongly dependent process, there must exist a set of weights satisfying (13)-(14) such that

$$\lim_{N \to \infty} Var(\bar{u}_{wt} | \mathcal{I}_{t-1}) = \bar{\gamma}_{w_{t-1},1}^2 + \dots + \bar{\gamma}_{w_{t-1},m}^2 > 0,$$
(32)

which implies that there must at least be one factor $f_{\ell t}$ such that its loadings satisfy the condition, $\lim_{N \to \infty} \bar{\gamma}_{w_{t-1},\ell} \neq 0.$ In the literature on factor models, it is quite common to impose conditions on the loadings or on

the eigenvalues of the conditional covariance matrix, Σ_{ut} , of $\mathbf{u}_t = (u_{1t}, ..., u_{Nt})'$ that constrain the

form of cross section dependence carried by the factor structure. For example, Bai (2005) assumes that the factor loadings satisfy $\sum_{i=1}^{N} \gamma_{i\ell}^2 = O(N)$, for $\ell = 1, ..., m$. But this condition does not necessarily imply $\lim_{N\to\infty} \bar{\gamma}_{w_{t-1},\ell} \neq 0$. It is, therefore, possible for a factor structure to exhibit weak cross section dependence in the sense of Definition 1. This, for example, could arise in the case where the factor loadings are independent draws from random a coefficient model with *zero means* and finite variances.

In the literature on asset pricing models, one common assumption is that $\lambda_m(\Sigma_{ut})$ is bounded away from zero at rate N (Chamberlain, 1983; Forni and Lippi, 2001). Under this assumption, since $rank(\Sigma_{ut}) = m$, and $\lambda_i(\Sigma_{ut}) > 0$, for i = 1, 2, ..., m, and $\lambda_i(\Sigma_{ut}) = 0$, for i = m + 1, m + 2, ..., N. Noting that under Assumption 7 $\lambda_1(\Sigma_{et}) = O(1)$, we have

$$\lambda_m\left(\mathbf{\Sigma}_t\right) \geq \lambda_m\left(\mathbf{\Sigma}_{ut}\right),\,$$

and

$$\lambda_{m+1}\left(\mathbf{\Sigma}_{t}
ight) \leq \lambda_{m+1}\left(\mathbf{\Sigma}_{ut}
ight) + \lambda_{1}\left(\mathbf{\Sigma}_{et}
ight) = \lambda_{1}\left(\mathbf{\Sigma}_{et}
ight).$$

Hence, it follows that $\lambda_1(\Sigma_t), ..., \lambda_m(\Sigma_t)$ increase without bound as $N \to \infty$, while $\lambda_{m+1}(\Sigma_t), ..., \lambda_N(\Sigma_t)$ satisfy the bounded eigenvalue condition. Such a model is generally referred to as an approximate factor model. Most factor structures yield eigenvalues that increase at rate N. But as shown by Kapetanios and Marcellino (2006), it is possible to devise factor models that generate eigenvalues that rise at rate N^d , for 0 < d < 1.

6 Weak cross section dependence in spatial models

When a metric of distance is available, spatial techniques can be used to allow for interactions and cross dependencies in regression model. Important applications of spatial models in economics can be found in the literature on regional income growth (Rivera-Batiz and Romer, 1991), in the field of microeconomics that analyses the diffusion processes of certain behaviours across a population (Brock and Durlauf, 2001), and in a recent strand of public economics that studies the role of interaction among policy makers (Brueckner, 2003).

In the space-time process $\{z_{it}, i \in S, t \in T\}$, the index *i* denotes the *i*th spatial unit whose location s_i is known and varies over a fixed subset of the *k*-dimensional Euclidean space \mathbb{R}^k . The spatial approach to cross section dependence assumes that the covariance between any two random variables depends on their relative distance. The ordering or neighbour relation among the variables $(z_{1t}, ..., z_{Nt})$ is expressed by means of the spatial weights matrix \mathbf{S}_t (Anselin, 1988, 2002). The specification of \mathbf{S}_t is in general arbitrary, typically based on some measures of distance between units. A range of suggestions have been offered in the literature, based on geographic distance (Cliff and Ord, 1981; Anselin, 1988), as well as more general metrics, such as the ones based on economic (Conley, 1999; Pesaran, Schuermann and Weiner, 2004), or social proximity (Conley and Topa, 2002). When dealing with geographic distance, it is common to define weights using the notion of contiguity between units, and assigning a nonzero weight only when units *i* and *j* are contiguous. There exists a number of different ways to build a matrix that contains contiguity information for the same spatial layout; the essential point is deciding whether units that only share a common vertex should be considered neighbours or not. In the example of a regular square lattice, these alternatives are denoted as the *rook* formation under which two units are considered as neighbours if they share common boundaries; the *bishop* formation, if they share common vertices, and the *queen* formation if they share both boundaries and vertices. Alternatively, weights can have continuous values, in general a declining function of the distance between spatial observations, such as $s_{ij,t} = e^{-\varphi d_{ij,t}}$, or $s_{ij,t} = 1/d_{ij,t}^{\varphi}$, where $d_{ij,t}$ is the distance between units at time *t*, and φ is a parameter that can be set a priori, or estimated jointly (Haining, 2003). Finally, some authors have suggested to use a block structure for the weighs matrix, by dividing the population into groups, and considering as neighbours only those units that belong to the same group (Case, 1991).

In this section we show that, under certain regularity conditions, cross section correlations arising from most widely applied spatial models can be considered as particular examples of CWD processes. We prove that, given certain conditions on the weights matrix, the largest eigenvalue of the covariance matrix of our sequence of variables is bounded as the number of cross section units is increased.

Consider the variables e_{it} for i = 1, ..., N collected in the vector $\mathbf{e}_t = (e_{1t}, ..., e_{Nt})'$, for t = 1, 2, ..., T. The process $\{e_{it}\}$ follows a Spatial Moving Average (SMA) process (Haining, 1978) if

$$\mathbf{e}_t = \delta \mathbf{S}_t \boldsymbol{\varepsilon}_t + \boldsymbol{\varepsilon}_t, \tag{33}$$

where δ is a scalar parameter and $\varepsilon_t = (\varepsilon_{1t}, ..., \varepsilon_{Nt})'$. In the spatial econometric literature where the focus is often on a single cross section, it is typically assumed that $\varepsilon_{it} \sim IID(0, \sigma^2)$. But in what follows we consider a less restrictive set up and allow the idiosyncratic errors, ε_{it} , to be serially correlated with mean zero and a finite variance $0 < \sigma_i^2 < K < \infty$. However, we continue to assume that ε_{it} and $\varepsilon_{jt'}$ are independently distributed for all t, t'; and for all $i \neq j$.

For the spatial weight matrix, \mathbf{S}_t , we make the following assumption:

ASSUMPTION 8: The main diagonal elements of \mathbf{S}_t are zero,

- (a) $\delta \in \left(-\frac{1}{\rho_{\mathbf{S}}}, \frac{1}{\rho_{\mathbf{S}}}\right)$, where $\rho_{\mathbf{S}} = \max_{1 \le t \le T} \left\{\rho\left(\mathbf{S}_{t}\right)\right\}$,
- (b) $\|\mathbf{S}_t\|_r$ is bounded for all t,
- (c) $\|\mathbf{S}_t\|_c$ is bounded for all t.

For the SMA model the covariance matrix of \mathbf{e}_t is

$$\mathbf{\Sigma}_t = \left(\mathbf{I}_N + \delta \mathbf{S}_t
ight) \mathbf{\Lambda}_arepsilon \left(\mathbf{I}_N + \delta \mathbf{S}_t'
ight),$$

where Λ_{ε} is a diagonal matrix with σ_i^2 on its diagonal. We note that the SMA specification induces a short-range correlation across section, since the only off-diagonal non-zero elements of

the covariance matrix are those corresponding to non-zero elements in \mathbf{S}_t . Given the properties of matrix norms, the largest eigenvalue of $\boldsymbol{\Sigma}_t$ satisfies the following inequality

$$\lambda_{1}\left(\boldsymbol{\Sigma}_{t}\right) = \lambda_{1}\left[\left(\mathbf{I}_{N} + \delta\mathbf{S}_{t}\right)\boldsymbol{\Lambda}_{\varepsilon}\left(\mathbf{I}_{N} + \delta\mathbf{S}_{t}'\right)\right] \leq \sigma_{\max}^{2}\left\|\mathbf{I}_{N} + \delta\mathbf{S}_{t}\right\|_{c}\left\|\mathbf{I}_{N} + \delta\mathbf{S}_{t}'\right\|_{c},$$

where $\sigma_{\max}^2 = \max_{1 \le i \le N} \{\sigma_i^2\} < K$. Hence, under Assumptions 8(b) and 8(c), e_{it} is a weakly cross sectionally correlated process. As we will see later on in this section, bounded row and column norms of the weights matrix is a very common assumption in the spatial econometric literature. Note that this assumption rules out the case where, for some t, \mathbf{S}_t has dominant units.

We say that $\{e_{it}\}$ follows a Spatial Error Component (SEC) process (Kelejian and Robinson, 1995) if

$$\mathbf{e}_t = \delta \mathbf{S}_t \boldsymbol{\psi}_t + \boldsymbol{\varepsilon}_t, \tag{34}$$

where ψ_t and ε_t are two uncorrelated vectors of random variables, with zero mean and diagonal variance matrices Λ_{ψ} and Λ_{ε} , with elements ω_i^2 and σ_i^2 , respectively. In this case \mathbf{e}_t has covariance matrix

$$\boldsymbol{\Sigma}_t = \delta^2 \mathbf{S}_t \boldsymbol{\Lambda}_{\psi} \mathbf{S}_t' + \boldsymbol{\Lambda}_{\varepsilon}.$$

The range of spatial correlation induced by the SEC is very similar to that in the SMA, since it is constrained to close neighbours, for which the off-diagonal elements of the weighting matrix are non-zero. The largest eigenvalue of Σ_t satisfies

$$\lambda_1\left(\mathbf{\Sigma}_t\right) \le \delta^2 \omega_{\max}^2 \left\|\mathbf{S}_t\right\|_c \left\|\mathbf{S}_t\right\|_r + \sigma_{\max}^2.$$

where $\omega_{\max}^2 = \max_{1 \le i \le N} \{\omega_i^2\} < K$. As for the SMA process, Assumptions 8(b) and 8(c) ensure that the SEC model is cross sectionally weakly dependent.

Consider now the Spatial Autoregressive (SAR) process introduced at the end of Section 2:

$$\mathbf{e}_t = \delta \mathbf{S}_t \mathbf{e}_t + \boldsymbol{\varepsilon}_t. \tag{35}$$

Under Assumption 8(a), this is an invertible process and can be expanded according to the geometric series (Horn and Johnson, 1985, p. 301)

$$\mathbf{e}_t = \sum_{h=0}^{\infty} \left(\delta \mathbf{S}_t
ight)^h \boldsymbol{\varepsilon}_t.$$

The covariance matrix of the SAR process is

$$\boldsymbol{\Sigma}_{t} = (\mathbf{I}_{N} - \delta \mathbf{S}_{t})^{-1} \boldsymbol{\Lambda}_{\varepsilon} (\mathbf{I}_{N} - \delta \mathbf{S}_{t}')^{-1} = \left[\sum_{h=0}^{\infty} (\delta \mathbf{S}_{t})^{h} \right] \boldsymbol{\Lambda}_{\varepsilon} \left[\sum_{h=0}^{\infty} (\delta \mathbf{S}_{t}')^{h} \right].$$

As Σ_t contains matrix powers and products of \mathbf{S}_t , even if \mathbf{S}_t contains few non-zero values, the covariance structure induced by the SAR model is not sparse. This argument has been used in spatial econometrics literature to claim that the SAR process describes a *global* pattern of spatial

correlation, since it links all units in the system to each other (Anselin, 2002). As a result, the cross section correlation carried by a Spatial Autoregressive representation can be much more extensive than that implied by a SMA or SEC processes. However, we note that the SAR model, under some regularity conditions, is weakly cross sectionally correlated. Indeed, the largest eigenvalue of Σ_t satisfies the following inequality

$$\begin{aligned} \lambda_{1}\left(\boldsymbol{\Sigma}_{t}\right) &\leq \sigma_{\max}^{2}\sum_{h=0}^{\infty}\left|\delta\right|^{h}\left\|\boldsymbol{\mathrm{S}}_{t}^{h}\right\|_{c}\sum_{h=0}^{\infty}\left|\delta\right|^{h}\left\|\boldsymbol{\mathrm{S}}_{t}^{h}\right\|r,\\ &\leq \sigma_{\max}^{2}\sum_{h=0}^{\infty}\left|\delta\right|^{h}\left\|\boldsymbol{\mathrm{S}}_{t}\right\|_{c}^{h}\sum_{h=0}^{\infty}\left|\delta\right|^{h}\left\|\boldsymbol{\mathrm{S}}_{t}\right\|_{r}^{h}.\end{aligned}$$

Hence, under Assumptions 8(b) and 8(c), and under the condition

$$\delta < 1/ \|\mathbf{S}_t\|_c$$
, and $\delta < 1/ \|\mathbf{S}_t\|_r$, (36)

we have

$$\lambda_1\left(\mathbf{\Sigma}_t\right) < \left\|\mathbf{\Sigma}_t\right\|_c \le \sigma_{\max}^2\left(\frac{1}{1-\left|\delta\right| \left\|\mathbf{S}_t\right\|_c}\right) \left(\frac{1}{1-\left|\delta\right| \left\|\mathbf{S}_t\right\|_r}\right) < K,$$

namely \mathbf{e}_t is a CWD process. Note that (36) also implies that the spatial autoregressive process is invertible, hence, under (36), Assumption 8(a) is not needed.

The conditions on bounded row and column norms of \mathbf{S}_t implies that each unit has a finite number of neighbours that does not increase with N, namely \mathbf{S}_t has no dominant units (columns/rows). This way of expanding the spatial weight matrix as the number of cross section units gets larger corresponds to the notion of increasing-domain asymptotics, which is the asymptotic based on a growing spatial domain, rather than on increasingly dense observations in a fixed and bounded region (Cressie, 1993).

We conclude this section noting that the SMA and the SAR processes, under the assumption of invertibility (i.e. Assumption 8(a) for the SAR process), are particular cases of the following general process

$$\mathbf{e}_t = \mathbf{R}_t \boldsymbol{\varepsilon}_t, \tag{37}$$

where \mathbf{R}_t is a given $N \times N$ matrix. Specifically, for a SMA process $\mathbf{R}_t = (\mathbf{I}_N + \delta \mathbf{S}_t)$, and for a SAR process $\mathbf{R}_t = (\mathbf{I}_N - \delta \mathbf{S}_t)^{-1}$. This matrix plays a crucial role in the analysis of spatial models, and as we shall see below typically satisfies the following assumption:

Assumption 9: \mathbf{R}_t has bounded row and column norms for all t.

The key result for the cross section averages of \mathbf{e}_t is summarized in the following theorem.

Theorem 17 Consider the process (37), where \mathbf{R}_t satisfies Assumption 9, and for each t, ε_{it} is distributed independently across i with mean zero, finite variances $0 < \sigma_i^2 < \sigma_{\max}^2 < \infty$, and finite fourth-order moments, $E(\varepsilon_{it}^4) = \mu'_{i4} < K < \infty$. Then for all sets of weights satisfying (13)-(14),

$$E\left(\bar{e}_{wt}^{2} | \mathcal{I}_{t-1}\right) = O(N^{-1}), and Var\left(\bar{e}_{wt}^{2} | \mathcal{I}_{t-1}\right) = O(N^{-2}).$$

$$E\left(\bar{e}_{wt}^{2} | \mathcal{I}_{t-1}\right) = \mathbf{w}_{t-1}^{\prime} \mathbf{R}_{t} \mathbf{\Lambda}_{\varepsilon} \mathbf{R}_{t}^{\prime} \mathbf{w}_{t-1} \leq \sigma_{\max}^{2} \left(\mathbf{w}_{t-1}^{\prime} \mathbf{w}_{t-1}\right) \lambda_{1} \left(\mathbf{R}_{t} \mathbf{R}_{t}^{\prime}\right).$$

But since \mathbf{R}_t has bounded row and column norms, $\lambda_1 (\mathbf{R}_t \mathbf{R}'_t)$ is bounded, and for all sets of weights satisfying (13)-(14) we have

$$E\left(\bar{e}_{wt}^2 \left| \mathcal{I}_{t-1} \right. \right) = O(N^{-1})$$

To prove the second part of the theorem, let

$$z_t = (\kappa'_t \kappa_t)^{-1/2} \kappa'_t \eta_t,$$

where

$$\boldsymbol{\kappa}_t = \boldsymbol{\Lambda}_{\varepsilon}^{1/2} \mathbf{R}_t' \mathbf{w}_{t-1}, \text{ and } \boldsymbol{\eta}_t = \boldsymbol{\Lambda}_{\varepsilon}^{-1/2} \boldsymbol{\varepsilon}_t.$$
(38)

Then

$$\bar{e}_{wt}^2 = \left(\mathbf{w}_{t-1}'\mathbf{R}_t\boldsymbol{\varepsilon}_t\boldsymbol{\varepsilon}_t'\mathbf{R}_t'\mathbf{w}_{t-1}\right)^2 = (\boldsymbol{\kappa}_t'\boldsymbol{\kappa}_t)z_t^2,$$

and

$$Var\left(\bar{e}_{wt}^{2} \left| \mathcal{I}_{t-1} \right. \right) = (\kappa_{t}' \kappa_{t})^{2} Var(z_{t}^{2}).$$

But for each t, z_t is a scalar random variable with mean zero, a unit variance, and finite fourth order moment (since it is a linear function of ε_t). Note also that since ε_{it} are cross sectionally independently distributed then by standard central limit theorems $z_t^2 \stackrel{a}{\backsim} \chi_1^2$ as $N \to \infty$. Hence $Var(z_t^2)$ is finite for all N. As a result

$$Var\left(\bar{e}_{wt}^{2} \left| \mathcal{I}_{t-1} \right. \right) \leq K(\boldsymbol{\kappa}_{t}' \boldsymbol{\kappa}_{t})^{2}$$

Now substituting for κ_t from (38) we have

$$Var\left(\bar{e}_{wt}^{2} \left| \mathcal{I}_{t-1} \right.\right) \leq \sigma_{\max}^{4} \left(\mathbf{w}_{t-1}^{\prime} \mathbf{w}_{t-1} \right)^{2} \left[\lambda_{1} \left(\mathbf{R}_{t} \mathbf{R}_{t}^{\prime} \right) \right]^{2}.$$

As $\lambda_1(\mathbf{R}_t\mathbf{R}'_t)$ and σ_{\max}^2 are bounded, then in view of (13)-(14), it follows that

$$Var\left(\bar{e}_{wt}^2 \left| \mathcal{I}_{t-1} \right. \right) = O(N^{-2}).$$

Therefore, for any invertible spatial process $\{e_{it}\}$, where the spatial weight matrix \mathbf{S}_t has bounded row and column norms, \bar{e}_{wt}^2 converges to zero in quadratic mean as $N \to \infty$, for all sets of predetermined weights satisfying the granularity conditions. It is worth emphasizing that this result holds even if the idiosyncratic errors, ε_{it} , follow heterogeneous processes that are serially correlated over time. But to identify the spatial model we would need some restrictions on their cross section correlations. For the purpose of proving the consistency of the CCE estimator it is sufficient that $\bar{e}_{wt}^2 \xrightarrow[]{q.m.}{\to} 0$ as $N \to \infty$. (see Section 7). The assumptions required for spatial correlation to be weak are often adopted in spatial econometric literature as regularity conditions, since they ensure consistency and asymptotic normality of maximum likelihood estimators (Lee, 2004; Kelejian and Prucha, 1999; Mardia and Marshall, 1984). Kelejian and Prucha (1999) prove consistency of their GMM estimator in the context of a single cross section under the assumption that the matrices **S** and $(\mathbf{I}_N - \delta \mathbf{S})^{-1}$, for $|\delta| < 1$, have bounded row and column norms. The proofs of consistency of GMM estimator (and also of ML) are based on a central limit theorem for quadratic forms in $\boldsymbol{\varepsilon}$ (see Kelejian and Prucha, 1999, 2001). Lee (2004) has studied the properties of ML and quasi-ML estimators of a spatial process where each unit is influenced by few neighboring units. The author shows that ML and quasi-ML estimators are consistent and asymptotic normal under the assumption that the matrix **S** has bounded row and column norms, and that $(\mathbf{I}_N - \delta \mathbf{S})^{-1}$ has bounded row and column norms at the true value of δ (say δ_0), and is uniformly bounded in either row or column sums for any other value of δ in a compact set around δ_0 .

Mardia and Marshall (1984) have demonstrated that in a regression equation with correlated errors, bounds on the largest eigenvalues of the error covariance matrix and of its first and second order derivatives (with respect to the unknown parameters) are sufficient for weak consistency and asymptotic normality of the ML estimator of the spatial parameters.

7 Estimating panels with common factors and spatial correlation

Consider the panel data model with common factors set out in equations (1)-(2), where e_{it} is an invertible spatial process of the type (37). From the discussion on strong and weak cross section correlation, it follows that Assumption 9 guarantees that (8) holds. Hence, the unobservable common factors can be well approximated by the cross section averages of the dependent variable and individual specific regressors. We now focus on estimation and inference on the means of the slope parameters β_i . To this end, we assume for ease of exposition that the rank condition (6) is satisfied, though this assumption can be relaxed. Let $\bar{\mathbf{M}}_w$ be defined by

$$\bar{\mathbf{M}}_w = \mathbf{I}_T - \bar{\mathbf{H}}_w (\bar{\mathbf{H}}'_w \bar{\mathbf{H}}_w)^{-1} \bar{\mathbf{H}}'_w$$

 $\bar{\mathbf{H}}_w = (\mathbf{D}, \bar{\mathbf{Z}}_w)$, where \mathbf{D} and $\bar{\mathbf{Z}}_w$ are, respectively, the matrices of observations on \mathbf{d}_t and $\bar{\mathbf{z}}_{wt} = (\bar{y}_{wt}, \bar{\mathbf{x}}_{wt})$. We make the following additional assumption:

ASSUMPTION 10: Consider the cross section averages of the individual specific variables, $\mathbf{z}_{it} = (y_{it}, \mathbf{x}'_{it})'$ defined by $\bar{\mathbf{z}}_{wt} = \sum_{i=1}^{N} w_i \mathbf{z}_{it}$, with time-invariant weights $\{w_j\}$ satisfying Assumption 5. Then the following conditions hold:

(a) The $k \times k$ matrices $\hat{\Psi}_{iT} = T^{-1}(\mathbf{X}'_i \mathbf{\bar{M}}_w \mathbf{X}_i)$ and $\hat{\Psi}_{ig} = T^{-1}(\mathbf{X}'_i \mathbf{M}_g \mathbf{X}_i)$, where $\mathbf{\bar{M}}_g = \mathbf{I}_T - \mathbf{G}(\mathbf{G}'\mathbf{G})^{-1}\mathbf{G}'$ and $\mathbf{G} = (\mathbf{D}, \mathbf{F})$, are non-singular and $\hat{\Psi}_{iT}^{-1}$ and $\hat{\Psi}_{ig}^{-1}$ have finite second order moments, for all i

(b) The matrix
$$\hat{\Psi}_{NT} = \sum_{i=1}^{N} w_i \frac{\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i}{T}$$
 is non-singular

Pesaran (2006) has suggested two alternative estimators for the means β of the individual slope coefficients. The Common Correlated Effects Mean Group (CCEMG) estimator and the Common Correlated Effects Pooled (CCEP) estimator. The latter is defined by

$$\hat{\mathbf{b}}_{MG} = N^{-1} \sum_{i=1}^{N} \hat{\mathbf{b}}_i,\tag{39}$$

where

$$\hat{\mathbf{b}}_i = (\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i)^{-1} \mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{y}_i, \tag{40}$$

and $\mathbf{X}_i = (\mathbf{x}_{i1}, ..., \mathbf{x}_{iT})', \mathbf{y}_i = (y_{i1}, ..., y_{iT})'$. The following theorem applies to this estimator.

Theorem 18 Consider the panel data model (1)-(2) with errors e_{it} given by (37). Suppose that Assumptions 1-5, 9 and 10(a) hold. Then the common correlated effects mean group estimator $\hat{\mathbf{b}}_{MG}$ given by (39) is asymptotically (for fixed T and $N \to \infty$) unbiased for $\boldsymbol{\beta}$, and as $(N,T) \xrightarrow{j} \infty$ we have

$$\sqrt{N}\left(\hat{\mathbf{b}}_{MG}-\boldsymbol{\beta}\right) \to N(0,\boldsymbol{\Sigma}_{MG}),$$

where

$$\boldsymbol{\Sigma}_{MG} = \boldsymbol{\Omega}_{\boldsymbol{v}}.$$

A proof is provided in the Appendix. The covariance matrix. Σ_{MG} , can be consistently estimated by

$$\hat{\boldsymbol{\Sigma}}_{MG} = \frac{1}{N-1} \sum_{i=1}^{N} \left(\hat{\mathbf{b}}_{i} - \hat{\mathbf{b}}_{MG} \right) \left(\hat{\mathbf{b}}_{i} - \hat{\mathbf{b}}_{MG} \right)'.$$
(41)

The CCEP estimator, which gains efficiency from pooling observations if $\beta_i = \beta$, is defined by

$$\hat{\mathbf{b}}_{P} = \left(\sum_{i=1}^{N} \mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{X}_{i}\right)^{-1} \sum_{i=1}^{N} \mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{y}_{i}.$$
(42)

The following theorem applies to this estimator.

Theorem 19 Consider the panel data model (1)-(2) with errors e_{it} given by (37). Suppose that Assumptions 1-5, 9 and 10(b) hold. Then the common correlated effects pooled estimator $\hat{\mathbf{b}}_P$ given by (42) is asymptotically unbiased for $\boldsymbol{\beta}$ and as $(N,T) \xrightarrow{j} \infty$ we have

$$\left(\sum_{i=1}^{N} w_i^2\right)^{-1/2} \left(\hat{\mathbf{b}}_P - \boldsymbol{\beta}\right) \to N(0, \boldsymbol{\Sigma}_P),$$

where

$$\boldsymbol{\Sigma}_P = \boldsymbol{\Psi}^{*-1} \mathbf{R}^* \boldsymbol{\Psi}^{*-1},$$

$$\begin{split} \Psi^* &= \lim_{N \to \infty} \left(\sum_{i=1}^N w_i \boldsymbol{\Sigma}_i \right), \\ \mathbf{R}^* &= \lim_{N \to \infty} \left[N^{-1} \sum_{i=1}^N \tilde{w}_i^2 \left(\boldsymbol{\Sigma}_i \boldsymbol{\Omega}_{\boldsymbol{v}} \boldsymbol{\Sigma}_i \right) \right], \\ \tilde{w}_i &= \frac{w_i}{\sqrt{N^{-1} \sum_{i=1}^N w_i^2}}. \end{split}$$

See the Appendix for a proof. A consistent estimator for Σ_P is (Pesaran, 2006)

$$\hat{\boldsymbol{\Sigma}}_P = \left(\sum_{i=1}^N w_i^2\right) \hat{\boldsymbol{\Psi}}^{*-1} \hat{\mathbf{R}}^* \hat{\boldsymbol{\Psi}}^{*-1},\tag{43}$$

with

$$\hat{\Psi}^* = \sum_{i=1}^N w_i \frac{\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i}{T},$$

$$\hat{\mathbf{R}}^* = \frac{1}{N-1} \sum_{i=1}^N \tilde{w}_i \frac{\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i}{T} \left\{ (\hat{\mathbf{b}}_i - \hat{\mathbf{b}}_{MG}) (\hat{\mathbf{b}}_i - \hat{\mathbf{b}}_{MG})' \right\} \frac{\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i}{T}$$

In the rest of the paper, we study the small sample properties of CCE estimators in a model with common factors and/or spatial correlations.

8 Monte Carlo experiments

8.1 Monte Carlo design

This section provides Monte Carlo evidence on the small sample properties of CCE estimators, under different assumptions on the stochastic process generating the error terms. The study is comprised of four sets of experiments. In the first set to be used as a benchmark, we replicate the Monte Carlo study in Pesaran (2006), where the dependent variable and the regressors are assumed to depend on a linear combination of unobserved common factors. In a second set of experiments, we consider a panel where the error terms are generated by a spatial autoregressive (SAR) process. In the third set, we combine the two sources of cross section dependence, assuming that the error term is the orthogonal sum of a factors structure and a spatial process. In the fourth set of experiments, we allow the source of cross section dependence to vary over time, assuming that in a certain interval of time the error term follows a factor structure, and in the remaining time periods is generated by a pure spatial process. For all experiments we considered the following data generating process

$$y_{it} = \alpha_i d_{1t} + \beta_{i1} x_{1it} + \beta_{i2} x_{2it} + \gamma_{i1} f_{1t} + \gamma_{i2} f_{2t} + e_{it},$$

$$x_{jit} = a_{ij1} d_{1t} + a_{ij2} d_{2t} + \gamma_{ij1} f_{1t} + \gamma_{ij2} f_{3t} + v_{ijt}, \quad j = 1, 2,$$

for i = 1, ..., N and t = 1, ..., T, where we assume homogeneous slopes with $\beta_i = \beta = (1, 1)'$. In the above equation, d_{1t} and d_{2t} are observed common factors, f_{1t} , f_{2t} , and f_{3t} are unobserved common effects, and e_{it} are idiosyncratic errors. We adopt the following data generating processes:

$$d_{1t} = 1, \quad d_{2t} = \delta_d d_{2t-1} + v_{dt}, \quad t = 1, ..., T,$$

$$v_{dt} \sim IIDN(0, 1 - \delta_d^2), \quad \delta_d = 0.5, \quad d_{2,0} = 0,$$

$$\begin{aligned} f_{jt} &= \delta_{f_j} f_{1t-1} + v_{f_{jt}}, \ j = 1, 2, 3; \ t = 1, ..., T, \\ v_{f_{jt}} &\sim IIDN(0, 1 - \delta_{f_j}^2), \ \delta_{f_j} = 0.5, \ f_{j0} = 0, \end{aligned}$$

and

$$\begin{aligned} v_{ijt} &= \delta_{v_{ijt}} v_{ijt-1} + v_{ijt}, \ t = 1, ..., T, \\ v_{ijt} &\sim IIDN(0, 1 - \delta_{v_{ijt}}^2), \ v_{ij0} = 0, \\ \delta_{v_{ijt}} &\sim IIDU(0.05, 0.95) \ \text{for} \ j = 1, 2. \end{aligned}$$

The factor loadings of the observed common effects do not change across replications and are generated as

$$\alpha_i \sim IIDN(1,1), \quad i = 1, ..., N,$$

 $(a_{i11}, a_{i21}, a_{i12}, a_{i22}) \sim IIDN(0.5\boldsymbol{\tau}_4, 0.5\mathbf{I}_4),$

where $\boldsymbol{\tau}_4 = (1, 1, 1, 1)'$ and \mathbf{I}_4 is a 4×4 identity matrix.

The various experiments involve different hypothesis on the loadings of the unobserved common factors, and on the way the errors e_{it} are generated. In particular, we have carried out the following sets of experiments:

Experiment A

The parameters of the unobserved common effects in the \mathbf{x}_{it} equation are generated as

$$\left(\begin{array}{ccc} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{array}\right) \sim IID \left(\begin{array}{ccc} N(0.5, 0.5) & 0 & N(0, 0.5) \\ N(0, 0.5) & 0 & N(0.5, 0.5) \end{array}\right),$$

the parameters of the unobserved common effects in the y_{it} equation are⁷

$$\gamma_{i1} \sim IIDN(1, 0.2), \ \gamma_{i2} \sim IIDN(1, 0.2).$$

Finally, in this experiment the errors are independently distributed across i, namely

$$e_{it} \sim IIDN(0, \sigma_i^2), \quad \sigma_i^2 \sim IIDU[0.5, 1.5], \text{ for } i = 1, ..., N_i$$

Experiment B

The factor loadings of the unobserved common effects are set to zero, $\gamma_{i11} = \gamma_{i13} = \gamma_{i21} = \gamma_{i23} = \gamma_{i1} = \gamma_{i2} = 0$. In other words, the regressors and the dependent variable do not depend on the unobserved factors \mathbf{f}_t . The individual specific errors e_{it} are generated according to a spatial autoregressive process

$$e_{it} = \delta \sum_{j=1}^{N} s_{ij} e_{jt} + \varepsilon_{it}, \text{ for } i = 1, ..., N, t = 1, ..., T$$
$$\varepsilon_{it} \sim IIDN(0, \sigma^2), \sigma^2 = 1,$$

where δ is the spatial autoregressive coefficient, s_{ij} , for i, j = 1, ..., N, are elements of a spatial weight matrix **S**, assumed to be time-invariant.

In this experiment we consider a spatial weight matrix **S** with elements $s_{ij} = 1$ if units *i* and *j* are adjacent and $s_{ij} = 0$ otherwise. Further, we assume that cross section units are ordered so that the p^{th} order neighbours of the i^{th} cross section unit can be defined as i - p and i + p cross section units. The spatial weight matrix is defined in a circular world, where the first observation is adjacent to the last observation. Finally the weighting matrix is row-standardized, so that all its non-zero elements are equal 1/2p. We note that such spatial weight matrix has bounded row and column sums.

We have experimented with two different values of the spatial coefficient, $\delta = 0.8$, which corresponds to a sizeable level of cross section error correlation, and $\delta = 0.4$ which correspond to a more moderate level of spatial error correlation. Further, we have experimented with a spatial weight matrix of order p = 2, and a sparser matrix of order p = 1. In total, Experiment B consists of four different sub-experiments.

This experiment is intended to illustrate the extent to which the inclusion of cross section averages of the dependent variable and regressors as in the CCE approach is able to capture spatial correlations.

⁷These parameters are drawn such that the rank condition (6) is satisfied.

Experiment C

The parameters of the unobserved common effects in the \mathbf{x}_{it} and in the y_{it} equations are generated as in Experiment A

$$\begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix} \sim IID \begin{pmatrix} N(0.5, 0.5) & 0 & N(0, 0.5) \\ N(0, 0.5) & 0 & N(0.5, 0.5) \end{pmatrix},$$

$$\gamma_{i1} \sim IIDN(1, 0.2), \gamma_{i2} \sim IIDN(1, 0.2).$$

The individual specific errors e_{it} are generated as in Experiment B, that is

$$e_{it} = \delta \sum_{j=1}^{N} s_{ij} e_{jt} + \varepsilon_{it}, \text{ for } i = 1, ..., N, t = 1, ..., T$$
$$\varepsilon_{it} \sim IIDN(0, \sigma^2), \sigma^2 = 1,$$

where δ is the spatial autoregressive coefficient, s_{ij} are the spatial weights.

Again, we have experimented the case of high spatial correlation $\delta = 0.8$, and the case of moderate spatial dependence $\delta = 0.4$, and changed the order of the weights matrix from p = 1 to p = 2. This set of experiments aims at verifying to what extent CCE estimators capture both local and global cross section dependence.

Experiment D

In these experiments the source of cross section dependence is allowed to vary over time. In particular, we assume that common factors have nonzero loadings in the equations for y_{it} and \mathbf{x}_{it} only over a certain interval of time, and assume that cross section dependence is generated by a spatial process for the remaining sample period. We consider the following two sub-sets of experiments, where the cross dependence changes from strong to weak and back to strong; and a second sub-set under which the cross dependence begins to be weak, turns into strong before reverting back to the weak form.

Experiment D1

• For $t = 1, ..., \lfloor \frac{T}{3} \rfloor$ parameters of the unobserved common effects in the \mathbf{x}_{it} and in the y_{it} equations are generated as in Experiment A

$$\begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix} \sim IID \begin{pmatrix} N(0.5, 0.5) & 0 & N(0, 0.5) \\ N(0, 0.5) & 0 & N(0.5, 0.5) \end{pmatrix},$$

$$\gamma_{i1} \sim IIDN(1, 0.2), \gamma_{i2} \sim IIDN(1, 0.2).$$

• $t = \lfloor \frac{T}{3} \rfloor + 1, ..., \lfloor \frac{2T}{3} \rfloor$, we set $\gamma_{i11} = \gamma_{i13} = \gamma_{i21} = \gamma_{i23} = \gamma_{i1} = \gamma_{i2} = 0$ (as in Experiment B),

and assume

$$e_{it} = \delta \sum_{j=1}^{N} s_{ij} e_{jt} + \varepsilon_{it}, \text{ for } i = 1, ..., N, t = 1, ..., T$$
$$\varepsilon_{it} \sim IIDN(0, \sigma^2), \sigma^2 = 1,$$

with $\delta = 0.4$.

• For $t = \lfloor \frac{2T}{3} \rfloor + 1, ..., T$ parameters of the unobserved common effects in the \mathbf{x}_{it} and in the y_{it} equations are generated as in Experiment A

$$\begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix} \sim IID \begin{pmatrix} N(0.5, 0.5) & 0 & N(0, 0.5) \\ N(0, 0.5) & 0 & N(0.5, 0.5) \end{pmatrix},$$

$$\gamma_{i1} \sim IIDN(1, 0.2), \gamma_{i2} \sim IIDN(1, 0.2).$$

Experiment D2

• For $t = 1, ..., \lfloor \frac{T}{3} \rfloor$ we set $\gamma_{i11} = \gamma_{i13} = \gamma_{i21} = \gamma_{i23} = \gamma_{i1} = \gamma_{i2} = 0$ (as in Experiment B), and assume

$$\begin{split} e_{it} &= & \delta \sum_{j=1}^{N} s_{ij} e_{jt} + \varepsilon_{it}, \quad \text{for } i = 1, ..., N, \ t = 1, ..., T \\ \varepsilon_{it} &\sim & IIDN(0, \sigma^2), \sigma^2 = 1, \end{split}$$

with $\delta=0.4$

• $t = \lfloor \frac{T}{3} \rfloor + 1, ..., \lfloor \frac{2T}{3} \rfloor$ the parameters of the unobserved common effects in the \mathbf{x}_{it} and in the y_{it} equations are generated as in Experiment A

$$\begin{pmatrix} \gamma_{i11} & 0 & \gamma_{i13} \\ \gamma_{i21} & 0 & \gamma_{i23} \end{pmatrix} \sim IID \begin{pmatrix} N(0.5, 0.5) & 0 & N(0, 0.5) \\ N(0, 0.5) & 0 & N(0.5, 0.5) \end{pmatrix},$$

$$\gamma_{i1} \sim IIDN(1, 0.2), \gamma_{i2} \sim IIDN(1, 0.2).$$

• For $t = \lfloor \frac{2T}{3} \rfloor + 1, ..., T$ we set $\gamma_{i11} = \gamma_{i13} = \gamma_{i21} = \gamma_{i23} = \gamma_{i1} = \gamma_{i2} = 0$ (as in Experiment B), and assume

$$\begin{split} e_{it} &= & \delta \sum_{j=1}^{N} s_{ij} e_{jt} + \varepsilon_{it}, \quad \text{for } i = 1, ..., N, \, t = 1, ..., T \\ \varepsilon_{it} &\sim & IIDN(0, \sigma^2), \sigma^2 = 1, \end{split}$$

with $\delta = 0.4$.

The aim of this set of experiments is to investigate the robustness of CCE estimators to the possible time variations in the nature of cross section dependence.

Each experiment was replicated 2,000 times for the (N, T) pairs with N, T = 20, 30, 50, 100, 200. In each experiment we computed the CCE Mean Group and the CCE Pooled estimator provided by formula (39) and (42), assuming equal weights $w_i = \frac{1}{N}$, i = 1, ..., N. We further considered a misspecified structure that ignores the presence of common factors and/or spatial correlations, i.e. the fixed effects estimator

$$\hat{\mathbf{b}}_{FE} = \left(\sum_{i=1}^{N} \mathbf{X}_{i}' \mathbf{M}_{\tau} \mathbf{X}_{i}\right)^{-1} \sum_{i=1}^{N} \mathbf{X}_{i}' \mathbf{M}_{\tau} \mathbf{y}_{i}, \tag{44}$$

where $\mathbf{M}_d = \mathbf{I}_T - \boldsymbol{\tau}(\boldsymbol{\tau}'\boldsymbol{\tau})^{-1}\boldsymbol{\tau}'$, and $\boldsymbol{\tau}$ is a vector of ones.

To facilitate the interpretation of results, in each experiment we computed a statistic of cross section dependence, the CD test (Pesaran, 2004), a statistic of local cross section correlation, the CD(p), and the simple average of pair-wise cross section correlation coefficients of the residuals, $\overline{\hat{r}}$. We have chosen these tests because they do not require the specification of a generating process for the error term. The CD statistic is

$$CD = \sqrt{\frac{2T}{N(N-1)}} \left(\sum_{i=1}^{N-1} \sum_{j=i+1}^{N} \hat{r}_{ij} \right),$$

where \hat{r}_{ij} is the sample estimate of the pair-wise correlation of the residuals, specifically

$$\hat{r}_{ij} = \frac{\sum_{t=1}^{T} \hat{u}_{it} \hat{u}_{jt}}{\left(\sum_{t=1}^{T} \hat{u}_{it}^2\right)^{1/2} \left(\sum_{t=1}^{T} \hat{u}_{jt}^2\right)^{1/2}},$$

and \hat{u}_{it} is an estimate of the regression residuals $u_{it} = y_{it} - \alpha_i d_{1t} - \beta' \mathbf{x}_{it}$, using the pooled estimator $\hat{\mathbf{b}}_P$ of β . Pesaran (2004) has shown that the CD test is suitable under global alternatives such as the multi-factor residual models. However, when the cross section units can be ordered, it is more appropriate to compute the following CD(p) test statistic

$$CD(p) = \sqrt{\frac{2T}{p(2N-p-1)}} \left(\sum_{s=1}^{p} \sum_{i=s+1}^{N} \hat{r}_{i,i-s} \right),$$

where p is the order of the spatial weight matrix. Finally, the average of pair-wise cross section correlation coefficients is

$$\bar{\hat{r}} = \frac{2}{N(N-1)} \left(\sum_{i=1}^{N-1} \sum_{j=i+1}^{N} \hat{r}_{ij} \right).$$

This Monte Carlo study is intended to investigate the relationship between the small sample properties of a number of estimators and the source of cross section dependence. In addition, this analysis provides interesting results for a number of issues. First, the performance of the fixed effects estimator in Experiment B highlights the consequences of ignoring spatial correlation in panels. Further, we can verify whether various assumptions of weak dependence in the error term have an impact on the asymptotic behaviour of the statistics of global and local cross section dependence. Finally, we can explore whether there is a link between the range and intensity of cross section dependence and the performance of estimators.

8.2 Monte Carlo results

Results on the estimation of the slope parameters for Experiments A, B, C and D are summarized in Tables 1, 2, 3, and 4, respectively. In what follows we focus on the estimation of β_1 ; results for β_2 are very similar and are not reported. Further, we only provide results for the spatial weight matrix of order p = 2, since a more sparse weights matrix (case p = 1) leads to very similar results. Finally, to save space, we only provide Size and Power of estimators for Experiments B and C, while we do not report their Bias and RMSE, since these are very close to Bias and RMSE values in Experiments A. Note that the power of the various tests are computed under the alternative $H_1: \beta_1 = 0.95$.

Table 1 shows results that are very similar to those in Pesaran (2006). First, the fixed effects estimator, ignoring the factor structure, performs very poorly, is substantially biased, and is subject to large size distortions. Further, as expected in the case of homogeneous slopes, the RMSE of the CCE Pooled estimators is lower than the RMSE of the CCE Mean Group estimator, although the difference between them becomes very small as N and T increase. Finally, tests using the CCE Mean Group and the CCE Pooled estimators are correctly sized, reflecting the fact that the estimated variance is a consistent estimate of the true variance.

Moving to Experiments B, Table 2 refers to the case where the error term is generated by a spatial autoregressive process. Monte Carlo results show that the fixed effects estimator, ignoring the spatial correlation, over-rejects the null hypothesis. The over-rejection tendency is due to the use of inappropriate standard errors, and appears to be substantial only in the case of the large value of the spatial parameter (namely when $\delta = 0.8$). Note that in the case of these experiments the FE estimator is consistent.

Focusing on the tests that use CCE Mean Group and CCE Pooled estimators, their empirical sizes are very close to the nominal size, for both values of the spatial parameter. Similarly, in Experiments C (Table 3), the combination of common factors and spatial correlation in the error term does not affect the empirical size of CCE estimators, which is close to the level of 5%. These Monte Carlo results clearly show that the CCE approach seems to work well not only in the case of global cross section dependence but also in the presence of spatial correlation. Finally, results from Experiments D (Table 4) suggest that CCE estimators are also robust to time variations in the source of cross section dependence.

Table 5 reports some statistics of cross section dependence for the above experiments, based on residuals from the CCE Pooled estimation. The average cross section correlation in Experiments A is high and persistent as N increases, for all values of T. Conversely, in Experiments B, \bar{r} declines rapidly to zero as the sample size is increased. In fact, local correlation implies that units have a limited number of neighbours regardless of the sample size. Therefore, the average of local cross section correlation computed over all units decreases as N increases. This also affects the CD statistic, which slightly diminishes as N is increased, for all values of T.

9 Concluding remarks

Cross section dependence is a rapidly growing field of study in panel data analysis. In the case of panels with large cross section dimension, unobserved factor models, or when a measure of distance among units is available, spatial processes are used. The essential aim of this paper has been to discuss the estimation of a panel regression model with common factors and spatial dependence. To this end, we introduced the notions of strong and weak cross section dependence, and show that these notions are critical to the distinction that prevails in the literature between the two approaches advanced for modelling of cross section dependence. An important extension of this paper would be to study of the properties of classical estimation techniques in regression models with cross sectionally weakly correlated errors. It might also be worth considering spatial patterns that are not weak and are dominated by one or more cross section units. This is likely to be a theoretically more meaningful approach to investigate common effects as compared to the purely statistical factor models the results of which are often difficult to interpret.

Appendix

We now sketch the proofs of Theorems 18 and 19. We refer to Pesaran (2006) for further details.

Lemma 20 Let $\mathbf{e}_t = \mathbf{R}_t \boldsymbol{\varepsilon}_t$, where \mathbf{R}_t satisfies Assumption 9, and for each t, $\boldsymbol{\varepsilon}_{it}$ is distributed independently across i and t with mean zero, finite variances $0 < \sigma_i^2 < K < \infty$, and finite fourthorder moments, $E(\boldsymbol{\varepsilon}_{it}^4) = \mu'_{i4} < K < \infty$. Then under Assumptions 1, and 2 we have for all sets of weights $\{w_i\}$ satisfying (13)-(14)

$$\frac{\mathbf{\bar{e}}'_{w}\mathbf{\bar{e}}_{w}}{T} = O_{p}\left(\frac{1}{N}\right), \tag{45}$$

$$\frac{\mathbf{F}'\bar{\mathbf{e}}_w}{T} = O_p\left(\frac{1}{\sqrt{NT}}\right), \frac{\mathbf{D}'\bar{\mathbf{e}}_w}{T} = O_p\left(\frac{1}{\sqrt{NT}}\right), \tag{46}$$

$$\frac{\mathbf{V}_i' \bar{\mathbf{e}}_w}{T} = O_p\left(\frac{1}{\sqrt{NT}}\right), \frac{\mathbf{e}_i' \bar{\mathbf{e}}_w}{T} = O_p\left(\frac{1}{\sqrt{NT}}\right), \tag{47}$$

where $\mathbf{\bar{e}}_w = (\bar{e}_{w1}, ..., \bar{e}_{wT})'$, $\bar{e}_{wt} = \sum_{i=1}^N w_i e_{it}$, \mathbf{D} and \mathbf{F} are $T \times n$ and $T \times m$ matrices on observed and unobserved common factors, and $\mathbf{V}_i = (\mathbf{v}_{i1}, ..., \mathbf{v}_{iT})'$.

Proof. Note that $T^{-1} \bar{\mathbf{e}}'_w \bar{\mathbf{e}}_w = T^{-1} \sum_{t=1}^T \bar{e}^2_{wt}$, by Theorem 17

$$E\left(T^{-1}\bar{\mathbf{e}}'_{w}\bar{\mathbf{e}}_{w}\right) = T^{-1}\sum_{t=1}^{T}E\left(\bar{e}_{wt}^{2}\right) = O\left(\frac{1}{N}\right).$$
(48)

Similarly, since ε_{it} are serially independent

$$Var\left(T^{-1}\mathbf{\bar{e}}'_{w}\mathbf{\bar{e}}_{w}\right) = T^{-2}\sum_{t=1}^{T} Var\left(\bar{e}_{wt}^{2}\right),$$

and by Theorem 17

$$Var\left(T^{-1}\mathbf{\bar{e}}'_{w}\mathbf{\bar{e}}_{w}\right) = O\left(\frac{1}{N^{2}T}\right).$$

which proves (45). As for (46), consider the ℓ th row of $T^{-1}\mathbf{F}'\bar{\mathbf{e}}_w$ and note that it can be written as $T^{-1}\sum_{t=1}^{T} f_{\ell t}\bar{e}_{wt}$, where $f_{\ell t}$ and \bar{e}_{wt} are distributed independently of each other. Then $T^{-1}\sum_{t=1}^{T} f_{\ell t}\bar{e}_{wt}$ has zero mean and variance

$$Var\left(T^{-1}\sum_{t=1}^{T} f_{\ell t}\bar{e}_{wt}\right) = T^{-2}\sum_{t=1}^{T} E\left(f_{\ell t}^{2}\right) E\left(\bar{e}_{wt}^{2}\right) \le \sigma_{\max}^{2} O\left(\frac{1}{N}\right) \left\{T^{-2}\sum_{t=1}^{T} E\left(f_{\ell t}^{2}\right)\right\} = O\left(\frac{1}{NT}\right).$$

This establishes (46). The second result in (46) and the first result in (47) follow similarly. As for the second result in (47), note that, since $w_i = O(1/N)$ and given that e_{it} is CWD

$$E(T^{-1}\mathbf{e}'_{i}\bar{\mathbf{e}}_{w}) = T^{-1}\sum_{t=1}^{T} E(e_{it}\bar{e}_{wt})$$

= $T^{-1}\sum_{t=1}^{T} [w_{1}E(e_{it}e_{1t}) + ... + w_{N}E(e_{it}e_{Nt})] = O\left(\frac{1}{N}\right)$

Denote the i^{th} row of \mathbf{R}_t by \mathbf{r}_{it} and note that⁸

$$\mathbf{r}_{it}'\mathbf{r}_{it} = \sum_{j=1}^{N} r_{ij,t}^2 \le \|\mathbf{r}_{it}\|_1^2 = O(1)$$

Hence we have

$$E\left(e_{it}^{4}\right) = E\left[\left(\mathbf{r}_{it}^{\prime}\boldsymbol{\varepsilon}_{t}\boldsymbol{\varepsilon}_{t}^{\prime}\mathbf{r}_{it}\right)^{2}\right] = E\left(\sum_{j=1}^{N}\sum_{\ell=1}^{N}r_{ij,t}r_{i\ell,t}\varepsilon_{jt}\varepsilon_{\ell t}\right)^{2}$$
$$= E\left(\sum_{j=1}^{N}\sum_{\ell=1}^{N}\sum_{k=1}^{N}\sum_{h=1}^{N}r_{ij,t}r_{i\ell,t}r_{ik,t}r_{ih,t}\varepsilon_{jt}\varepsilon_{\ell t}\varepsilon_{kt}\varepsilon_{ht}\right)$$
$$= \sum_{j=1}^{N}\sum_{k=1}^{N}r_{ij,t}^{2}r_{ik,t}^{2}E\left(\varepsilon_{jt}^{2}\varepsilon_{kt}^{2}\right) \leq K\left(\sum_{j=1}^{N}r_{ij,t}^{2}\right)^{2} = O(1),$$

and the second moment of $T^{-1}\mathbf{e}'_i \mathbf{\bar{e}}_w$ satisfies (recall that $E\left(\bar{e}^4_{wt}\right) = O(N^{-2})$)

$$E\left[\left(T^{-1}\mathbf{e}_{i}'\bar{\mathbf{e}}_{w}\right)^{2}\right] = T^{-2}\sum_{t=1}^{T}E\left(e_{it}^{2}\bar{e}_{wt}^{2}\right) \leq T^{-2}\sum_{t=1}^{T}\left[E\left(e_{it}^{4}\right)E\left(\bar{e}_{wt}^{4}\right)\right]^{1/2}$$
$$= O\left(\frac{1}{N}\right)T^{-2}\sum_{t=1}^{T}\left[E\left(e_{it}^{4}\right)\right]^{1/2} = O\left(\frac{1}{NT}\right).$$

The above results can be used to prove further results that are helpful in deriving the asymptotic distribution of CCE estimators. Let

$$oldsymbol{\xi}_{it} = \left(egin{array}{c} e_{it} + oldsymbol{eta}_i' \mathbf{v}_{it} \ \mathbf{v}_{it} \end{array}
ight)$$

•

⁸Let $\mathbf{x} = (x_1, ..., x_n)'$, we define the vector norm $\|\cdot\|_1$ of \mathbf{x} as $\|\mathbf{x}\|_1 = |x_1| + ... + |x_N|$.

From Lemma 20 it follows that (see also Lemma 2 and 3 in Pesaran (2006))

$$\begin{split} & \frac{\bar{\boldsymbol{\xi}}'_w \bar{\boldsymbol{\xi}}_w}{T} &= O_p \left(\frac{1}{N} \right), \\ & \frac{\mathbf{F}' \bar{\boldsymbol{\xi}}_w}{T} &= O_p \left(\frac{1}{\sqrt{NT}} \right), \ \frac{\mathbf{D}' \bar{\boldsymbol{\xi}}_w}{T} = O_p \left(\frac{1}{\sqrt{NT}} \right), \\ & \frac{\mathbf{V}'_i \bar{\boldsymbol{\xi}}_w}{T} &= O_p \left(\frac{1}{N} \right) + O_p \left(\frac{1}{\sqrt{NT}} \right), \\ & \frac{\mathbf{E}'_i \bar{\boldsymbol{\xi}}_w}{T} = O_p \left(\frac{1}{N} \right) + O_p \left(\frac{1}{\sqrt{NT}} \right), \\ & \frac{\mathbf{E}'_i \bar{\boldsymbol{\xi}}_w}{T} &= O_p \left(\frac{1}{N} \right) + O_p \left(\frac{1}{\sqrt{NT}} \right). \end{split}$$

The above results prove the fundamental relations (under the assumption that the rank condition is satisfied)

$$\frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{F}}{T} = O_{p} \left(\frac{1}{N} \right) + O_{p} \left(\frac{1}{\sqrt{NT}} \right), \tag{49}$$

$$\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{w}\mathbf{X}_{i}}{T} = \frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{g}\mathbf{X}_{i}}{T} + O_{p}\left(\frac{1}{N}\right) + O_{p}\left(\frac{1}{\sqrt{NT}}\right),$$
(50)

$$\frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{e}_{i}}{T} = \frac{\mathbf{X}_{i}' \bar{\mathbf{M}}_{g} \mathbf{e}_{i}}{T} + O_{p} \left(\frac{1}{N}\right), \tag{51}$$

where $\bar{\mathbf{M}}_g = \mathbf{I}_T - \mathbf{G}(\mathbf{G}'\mathbf{G})^{-1}\mathbf{G}'$. Note that (49)-(51) are identical to relations (40), (43) and (44) in Pesaran (2006), and will be used to derive the asymptotic distribution of CCE Mean Group and CCE Pooled estimators.

Proof of Theorem 18. Consider

$$\sqrt{N}\left(\mathbf{b}_{MG}-\boldsymbol{\beta}\right) = \frac{1}{\sqrt{N}}\sum_{i=1}^{N}\boldsymbol{\upsilon}_{i} + \frac{1}{N}\sum_{i=1}^{N}\hat{\boldsymbol{\Psi}}_{iT}^{-1}\left(\frac{\sqrt{N}\mathbf{X}_{i}'\bar{\mathbf{M}}_{w}\mathbf{F}}{T}\right)\boldsymbol{\gamma}_{i} + \frac{1}{N}\sum_{i=1}^{N}\hat{\boldsymbol{\Psi}}_{iT}^{-1}\left(\frac{\sqrt{N}\mathbf{X}_{i}'\bar{\mathbf{M}}_{w}\mathbf{e}_{i}}{T}\right),$$

where by assumption $\hat{\Psi}_{iT}^{-1} = (T^{-1}\mathbf{X}'_i \bar{\mathbf{M}}_w \mathbf{X}_i)^{-1}$ has second-order moments. Given (49) it follows that

$$\frac{\sqrt{N}\mathbf{X}_{i}'\bar{\mathbf{M}}_{w}\mathbf{F}}{T} = O_{p}\left(\frac{1}{\sqrt{N}}\right) + O_{p}\left(\frac{1}{\sqrt{T}}\right),\tag{52}$$

and hence, since by assumption factor loadings are bounded, we have (Pesaran, 2006, p. 983)

$$\frac{1}{N} \sum_{i=1}^{N} \hat{\boldsymbol{\Psi}}_{iT}^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{F}}{T} \right) \boldsymbol{\gamma}_{i} \xrightarrow{p} \mathbf{0}, \quad \text{as } (N,T) \xrightarrow{j} \infty.$$

From (50)-(51) we have

$$\frac{1}{N} \sum_{i=1}^{N} \hat{\boldsymbol{\Psi}}_{iT}^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \bar{\mathbf{M}}_{w} \mathbf{e}_{i}}{T} \right) = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \mathbf{M}_{g} \mathbf{X}_{i}}{T} \right)^{-1} \left(\frac{\sqrt{N} \mathbf{X}_{i}' \mathbf{M}_{g} \mathbf{e}_{i}}{T} \right) + O_{p} \left(\frac{1}{\sqrt{N}} \right) + O_{p} \left(\frac{1}{\sqrt{T}} \right)$$
$$= \boldsymbol{\Delta}_{NT} + O_{p} \left(\frac{1}{\sqrt{N}} \right) + O_{p} \left(\frac{1}{\sqrt{T}} \right),$$

where

$$\begin{aligned} \boldsymbol{\Delta}_{NT} &= \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \left(\frac{\mathbf{X}_{i}' \mathbf{M}_{g} \mathbf{X}_{i}}{T} \right)^{-1} \left(\frac{\mathbf{X}_{i}' \mathbf{M}_{g} \mathbf{e}_{i}}{T} \right) \\ &= \frac{1}{\sqrt{NT}} \sum_{i=1}^{N} \left(\frac{\mathbf{V}_{i}' \mathbf{M}_{g} \mathbf{V}_{i}}{T} \right)^{-1} \left(\frac{\mathbf{V}_{i}' \mathbf{M}_{g} \mathbf{e}_{i}}{\sqrt{T}} \right). \end{aligned}$$

 Δ_{NT} has zero mean and variance

$$Var(\mathbf{\Delta}_{NT}) = \frac{1}{NT} \sum_{i=1}^{N} \sum_{j=1}^{N} E\left[\left(\frac{\mathbf{V}_{i}'\mathbf{M}_{g}\mathbf{V}_{i}}{T}\right)^{-1} \left(\frac{\mathbf{V}_{i}\mathbf{M}_{g}\mathbf{e}_{i}\mathbf{e}_{j}'\mathbf{M}_{g}\mathbf{V}_{j}}{T}\right) \left(\frac{\mathbf{V}_{j}'\mathbf{M}_{g}\mathbf{V}_{j}}{T}\right)^{-1}\right]$$
$$= \frac{1}{NT} \sum_{i=1}^{N} \sum_{j=1}^{N} \mathbf{\Sigma}_{i}^{-1} E\left(\frac{\mathbf{V}_{i}'\mathbf{M}_{g}\mathbf{e}_{i}\mathbf{e}_{j}'\mathbf{M}_{g}\mathbf{V}_{j}}{T}\right) \mathbf{\Sigma}_{j}^{-1}.$$

Consider the $(t,s)^{th}$ element of the $T \times T$ matrix $E\left(\mathbf{e}_{i}\mathbf{e}_{j}'\right)$ and note that

$$E\left(e_{it}e_{js}\right) = E\left(\mathbf{r}'_{it}\boldsymbol{\varepsilon}_{t}\boldsymbol{\varepsilon}'_{s}\mathbf{r}_{jt}\right) = \begin{cases} \mathbf{r}'_{it}\mathbf{\Lambda}_{\varepsilon}\mathbf{r}_{jt}, & \text{for } t = s \\ 0, & \text{otherwise} \end{cases}$$

where \mathbf{r}_{it} is the *i*th row of the matrix \mathbf{R}_t . Given that the matrix \mathbf{R}_t has bounded absolute column and row sums, it follows that the diagonal elements of the matrix $E\left(\mathbf{e}_i\mathbf{e}_j'\right)$ satisfy

$$\mathbf{r}_{it}' \mathbf{\Lambda}_{\varepsilon} \mathbf{r}_{jt} \le \sigma_{\max}^2 \|\mathbf{r}_{it}\|_1 \|\mathbf{r}_{jt}\|_1 = O(1), \quad \text{for } t = 1, 2, ..., T.$$

Hence, since \mathbf{V}_i are independent across i, we have

$$Var(\mathbf{\Delta}_{NT}) \leq \frac{1}{NT} K \sum_{i=1}^{N} \sum_{j=1}^{N} \mathbf{\Sigma}_{i}^{-1} E\left(\frac{\mathbf{V}_{i}' \mathbf{M}_{g} \mathbf{V}_{j}}{T}\right) \mathbf{\Sigma}_{j}^{-1} = \frac{1}{NT} K \sum_{i=1}^{N} \mathbf{\Sigma}_{i}^{-1} = O\left(\frac{1}{T}\right).$$

Therefore

$$\sqrt{N} \left(\mathbf{b}_{MG} - \boldsymbol{\beta} \right) = \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \boldsymbol{v}_i + O_p \left(\frac{1}{\sqrt{N}} \right) + O_p \left(\frac{1}{\sqrt{T}} \right),$$

and

$$\sqrt{N} \left(\mathbf{b}_{MG} - \boldsymbol{\beta} \right) \xrightarrow{d} N \left(0, \boldsymbol{\Sigma}_{MG} \right) \quad \text{as } (N, T) \xrightarrow{j} \infty.$$

Proof of Theorem 19. Consider

$$\left(\sum_{i=1}^{N} w_i^2\right)^{-1/2} \left(\hat{\mathbf{b}}_P - \boldsymbol{\beta}\right) = \left(\sum_{i=1}^{N} w_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_w \mathbf{X}_i}{T}\right)^{-1} \left(\frac{1}{\sqrt{N}} \sum_{i=1}^{N} \tilde{w}_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_w \left(\mathbf{X}_i \boldsymbol{v}_i + \mathbf{e}_i\right)}{T} + \mathbf{q}_{NT}\right),$$

where

$$\tilde{w}_{i} = \left(N^{-1}\sum_{i=1}^{N}w_{i}^{2}\right)^{-1/2}w_{i} = O(1),$$
$$\mathbf{q}_{NT} = \frac{1}{\sqrt{N}}\sum_{i=1}^{N}\tilde{w}_{i}\frac{\mathbf{X}_{i}'\bar{\mathbf{M}}_{w}\mathbf{F}\boldsymbol{\gamma}_{i}}{T}.$$

Using (49) and (52)

$$\mathbf{q}_{NT} = \frac{1}{N} \sum_{i=1}^{N} \tilde{w}_i \frac{\sqrt{N} \mathbf{X}'_i \mathbf{\bar{M}}_w \mathbf{F} \boldsymbol{\gamma}_i}{T} \xrightarrow{p} \mathbf{0} \quad \text{as } (N,T) \xrightarrow{j} \infty.$$

Hence,

$$\left(\sum_{i=1}^{N} w_i^2\right)^{-1/2} \left(\hat{\mathbf{b}}_P - \boldsymbol{\beta}\right) = \left(\sum_{i=1}^{N} w_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_g \mathbf{X}_i}{T}\right)^{-1} \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \tilde{w}_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_g \left(\mathbf{X}_i \boldsymbol{v}_i + \mathbf{e}_i\right)}{T} + O_p \left(\frac{1}{\sqrt{N}}\right) + O_p \left(\frac{1}{\sqrt{T}}\right).$$

Note that the element

$$\left(\sum_{i=1}^{N} w_i \frac{\mathbf{X}'_i \bar{\mathbf{M}}_g \mathbf{X}_i}{T}\right)^{-1} \frac{1}{\sqrt{N}} \sum_{i=1}^{N} \frac{\mathbf{X}'_i \bar{\mathbf{M}}_g \mathbf{e}_i}{T}$$

has mean zero and variance

$$\left(\sum_{i=1}^{N} w_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_g \mathbf{X}_i}{T}\right)^{-1} \frac{1}{N} \sum_{i=1}^{N} \sum_{j=1}^{N} \tilde{w}_i \tilde{w}_j E\left(\frac{\mathbf{V}_i' \bar{\mathbf{M}}_g \mathbf{e}_i \mathbf{e}_j' \bar{\mathbf{M}}_g \mathbf{V}_j}{T}\right) \left(\sum_{i=1}^{N} w_i \frac{\mathbf{X}_i' \bar{\mathbf{M}}_g \mathbf{X}_i}{T}\right)^{-1}$$
$$= O\left(\frac{1}{T}\right),$$

considering that $E\left(\mathbf{e}_{i}\mathbf{e}_{j}'\right)$ is a diagonal matrix with finite elements, and \mathbf{V}_{i} is independently distributed across *i*. It then follows that

$$\left(\sum_{i=1}^{N} w_i^2\right)^{-1/2} \left(\hat{\mathbf{b}}_P - \boldsymbol{\beta}\right) \to N(0, \boldsymbol{\Sigma}_P).$$

References

- Anselin, L. (1988), Spatial Econometrics: Methods and Models, Kluwer Academic Publishers, Dordrecht, the Netherlands.
- [2] Anselin, L. (2002), Under the hood. Issues in the specification and interpretation of spatial regression models, Agricultural Economics, 27, 247-267.
- [3] Bai, J., and Ng, S. (2004), A PANIC Attack on Unit Roots and Cointegration, *Econometrica*, 72, 1127-1177.
- Bai, J. (2005), Panel Data Models with Interactive Fixed Effects, Unpublished Manuscript, New York University.
- [5] Bernstein, D.S. (2005), *Matrix Mathematics*, Princeton University Press.
- [6] Breitung, J., and Pesaran M.H. (2006), Unit Roots and Cointegration in Panels, in L. Matyas and Sevestre P. (eds.) The Econometrics of Panel Data: Fundamentals and Recent Developments in Theory and Practice, Kluwer Academic Publishers, forthcoming.
- [7] Brock, W., and Durlauf, S. (2001), Interactions-Based Models, Handbook of Econometrics, Volume V, Heckman, J., Leamer, E., (eds), North Holland, Amsterdam.
- [8] Brueckner, J.K. (2003), Strategic Interaction Among Governments: an Overview of Empirical Studies, International Regional Science Review, 26, 175-188.
- [9] Case, A.C. (1991), Spatial Pattern in Household Demand, Econometrica, 59, 953-965.
- [10] Chamberlain, G. (1983), Funds, Factors, and Diversification in Arbitrage Pricing Models, *Econometrica*, 51, 1281-1304.
- [11] Chamberlain, G., and Rothschild, M. (1983), Arbitrage, Factor Structure and Mean-variance Analysis in Large Asset Market, *Econometrica*, 51, 1305-1324.
- [12] Chang, Y. (2004), Bootstrap Unit Root Test in Panels with Cross-Sectional Dependency, Journal of Econometrics, 120, 263-293.
- [13] Choi, I. (2002), Combination Unit Root Tests for Cross-sectionally Correlated Panels, in Econometric Theory and Practice: Frontiers of Analysis and Applied Research, Essays in Honor of Peter C.,B., Phillips, Cambridge University Press.
- [14] Cliff, A., and Ord, J.K. (1973), Spatial Autocorrelation, Pion, London.
- [15] Cliff, A., and Ord, J.K. (1981), Spatial Processes: Models and Applications, Pion, London.
- [16] Coakley, J., Fuertes, A.M., and Smith, R. (2002), A Principal Components Approach to Cross-Section Dependence in Panels, Birkbeck College Discussion Paper 01/2002.

- [17] Coakley, J., Fuertes, A.M., and Smith, R. (2006), Unobserved Heterogeneity in Panel Time Series., Computational Statistics and Data Analysis, 50, 2361-2380.
- [18] Conley, T.G. (1999), GMM Estimation with Cross Sectional Dependence, Journal of Econometrics, 92, 1-45.
- [19] Conley, T.G., and Topa, G. (2002), Socio-economic Distance and Spatial Patterns in Unemployment, Journal of Applied Econometrics, 17, 303-327.
- [20] Connor, G., and Korajzcyk, R.A. (1993), A Test for the Number of Factors in an Approximate Factor Structure, *The Journal of Finance*, 48, 1263-1291.
- [21] Cressie, N. (1993), Statistics for Spatial Data, Wiley, New York.
- [22] Forni, M., and Lippi, M. (2001), The Generalized Factor Model: Representation Theory, Econometric Theory, 17, 1113-1141.
- [23] Forni, M., and Reichlin, L. (1998), Let's Get Real: a Factor Analytical Approach to Disaggregated Business Cycle Dynamics, *Review of Economic Studies*, 65, 453-473.
- [24] Gregory, A., and Head, A. (1999), Common and Country Specific Fluctuations in Productivity, Investment, and the Current Account, *Journal of Monetary Economics*, 44, 423-452.
- [25] Haining, R.P. (2003), Spatial Data Analysis: Theory and Practice, University Press, Cambridge.
- [26] Haining, R.P. (1978), The Moving Average Model for Spatial Interaction, Transactions of the Institute of British Geographers, 3, 202-225.
- [27] Horn, R.A., and Johnson, C.A. (1985), *Matrix Analysis*, Cambridge University Press.
- [28] Im, K.S., Pesaran, M.H., and Shin, Y. (2003), Testing for Unit Roots in Heterogeneous Panels, Journal of Econometrics, 115, 53-74.
- [29] Kapetanios, G. and Marcellino, M. (2006), Factor-GMM Estimation with Large Sets of Possibly Weak Instruments, Unpublished Manuscript, Queen Mary College, University of London.
- [30] Kapetanios, G., and Pesaran, M.H. (2007), Alternative Approaches to Estimation and Inference in Large Multifactor Panels: Small Sample Results with an Application to Modelling of Asset Returns, in Phillips, G., Tzavalis, E. (Eds.,), *The Refinement of Econometric Estimation and Test Procedures: Finite Sample and Asymptotic Analysis*, Cambridge University Press, Cambridge.
- [31] Kapetanios, G., Pesaran, M.H., and Yagamata, T. (2006), Panels with Nonstationary Multifactor Error Structures, CESifo Working Paper No. 1788.
- [32] Kelejian, H.H., and Prucha, I. (2001), On the Asymptotic Distribution of the Moran I Test Statistic with Applications, *Journal of Econometrics*, 104, 219-257.
- [33] Kelejian, H.H., and Prucha, I. (1999), A Generalized Moments Estimator for the Autoregressive Parameter in a Spatial Models, International Economic Review, 40, 509-533.

- [34] Kelejian, H.H., and Robinson, D.P. (1995), Spatial correlation: A suggested Alternative to the Autoregressive Model, in Anselin, L., and Florax, R.J. (eds), New Directions in Spatial Econometrics, 75-95, Springer-Verlag, Berlin.
- [35] Kelejian, H., Kapoor, M., and Prucha, I. (2006), Panel Data Models with Spatially Correlated Error Components, Journal of Econometrics, in Press.
- [36] Lee, L.F. (2002), Consistency and Efficiency of Least Squares Estimation for Mixed Regressive, Spatial Autoregressive Models, *Econometric Theory*, 18, 252-277.
- [37] Lee, L.F. (2004), Asymptotic Distributions of Quasi-Maximum Likelihood Estimators for Spatial Autoregressive Models, *Econometrica*, 72, 1899-1925.
- [38] Levin, A., Lin, C.F., and Chu, J. (2002), Unit Root Tests in Panel Data: Asymptotic and Finite Sample Properties, *Journal of Econometrics*, 108, 1-24.
- [39] Mardia, K.V., and Marshall, R.J. (1984), Maximum Likelihood Estimation of Models for Residual Covariance in Spatial Regression, *Biometrika*, 71, 135-146.
- [40] Moon, H.R., and Perron B. (2004), Testing for a Unit Root in Panels with Dynamic Factors, Journal of Econometrics, 122, 81-126.
- [41] O'Connell, P.G.J. (1998), The Overvaluation of Purchasing Power Parity, Journal of International Economics, 44, 1-19.
- [42] Pesaran, M.H. (2004), General Diagnostic Tests for Cross Section Dependence in Panels, CESifo Working Paper Series n. 1229, IZA Discussion Paper No. 1240.
- [43] Pesaran, M.H. (2006), Estimation and Inference in Large Heterogenous Panels with Multifactor Error Structure, *Econometrica*, 74, 967-1012.
- [44] Pesaran, M.H. (2007), A Simple Panel Unit Root Test in the Presence of Cross Section Dependence, Journal of Applied Econometrics, 22, 265-312.
- [45] Pesaran, M.H., and Chudik, A. (2007), Infinite Dimensional VARs, Paper under preparation, University of Cambridge.
- [46] Pesaran, M.,H., Schuermann, T., and Weiner, S. (2004), Modelling Regional Interdependencies using a Global Error-Correcting Macroeconometric Model, *Journal of Business and Economics Statistics*, 22, 129-162.
- [47] Phillips, P.C.B., and Sul, D. (2003), Dynamic Panel Estimation and Homogeneity Testing Under Cross Section Dependence, *The Econometrics Journal*, 6, 217-259.
- [48] Pinkse, J., Slade, M.E., and Brett, C. (2002), Spatial Price Competition, Econometrica, 70, 1111-1153.
- [49] Rivera-Batiz, L.A., and Romer, P.M. (1991), Economic Integration and Endogenous Growth, The Quarterly Journal of Economics, 106, 531-555.

- [50] Robertson, D., and Symons, J. (2000), Factor Residuals in SUR Regressions: Estimating Panels allowing for Cross Sectional Correlation, Unpublished Manuscript, Faculty of Economics and Politics, University of Cambridge.
- [51] Robinson, P.M. (2003), Long Memory Time Series. In *Time Series with Long Memory*. Edited by Robinson, P. M. Oxford University Press, pp. 4-32.
- [52] Spearman, C. (1904), General Intelligence Objectively Determined and Measured, American Journal of Psychology, 15, 201-293.
- [53] Stock, J.H., and Watson, M.W. (2002), Macroeconomic Forecasting Using Diffusion Indexes, Journal of Business & Economic Statistics, 20, 147-162.
- [54] Stone, R.N. (1947), On the Interdependence of Blocks of Transactions, Supplement to the Journal of the Royal Statistical Society, 9, 1-45.
- [55] Whittle, P. (1954), On Stationary Processes on the Plane, Biometrika, 41, 434-449.
- [56] Zellner, A. (1962), An Efficient Method of Estimating Seemingly Unrelated Regressions and Tests for Aggregation Bias, *Journal of the American Statistical Association*, 57, 348-368.

			В	$sias (\times 10)$	0)				RMSE	$(\times 100)$		
				Т					r	Г		
N	10	20	30	50	100	200	10	20	30	50	100	200
						CCE Me	an Group	þ				
20	0.68	0.00	-0.07	-0.08	0.04	0.02	21.54	8.15	5.86	4.29	2.96	2.21
30	0.12	-0.05	0.05	0.03	-0.02	0.04	18.40	6.69	4.81	3.30	2.22	1.66
50	-0.26	0.00	0.12	-0.02	0.03	0.06	14.35	5.17	3.61	2.51	1.62	1.17
100	-0.19	-0.02	0.02	-0.02	0.02	0.01	10.39	3.65	2.56	1.77	1.11	0.74
200	0.13	0.06	-0.00	-0.03	0.01	-0.00	7.55	2.70	1.85	1.21	0.78	0.53
						CCE	Pooled					
20	0.19	-0.01	-0.02	-0.06	-0.00	0.03	12.06	6.92	5.14	4.03	2.96	2.28
30	0.05	-0.06	-0.04	0.06	-0.03	0.03	10.14	5.48	4.15	3.06	2.16	1.68
50	-0.36	-0.02	0.09	-0.04	0.03	0.07	7.91	4.24	3.15	2.26	1.58	1.16
100	0.02	0.06	0.07	-0.01	0.01	0.01	5.70	2.95	2.23	1.63	1.06	0.72
200	0.09	0.02	0.006	-0.01	0.01	-0.00	4.01	2.09	1.59	1.11	0.74	0.52
					Fix	ed Effec	ts Estima	ator				
20	17.43	17.23	17.13	17.52	17.20	17.48	23.44	20.98	20.03	19.62	18.70	18.67
30	16.62	17.00	17.08	17.41	17.55	17.51	23.34	20.90	20.03	19.36	18.83	18.53
50	17.50	18.14	18.41	18.34	18.33	18.10	21.65	20.62	20.17	19.67	19.17	18.75
100	16.57	17.26	17.09	17.59	17.25	17.11	21.08	19.86	18.90	18.79	17.95	17.56
200	16.33	17.24	17.28	17.35	17.37	17.35	20.31	19.65	18.91	18.41	17.97	17.69

Table 1: Small Sample Properties of CCE Type Estimators in Experiments A

		Size	(5% leve)	el) (×100	$(H_0: \beta)$	$_1 = 1)$	Po	wer (5%)	evel (×	$(100)(H_1)$	$: \beta_1 = 0.$	95)
				Т						Т		
Ν	10	20	30	50	100	200	10	20	30	50	100	200
						CCE M	ean Grou	ıp				
20	5.90	6.45	6.45	8.25	7.25	7.25	6.65	12.50	17.25	26.85	45.90	67.30
30	6.05	6.25	6.95	6.85	5.85	6.80	7.15	13.15	20.80	35.70	62.95	87.15
50	4.80	6.15	5.50	5.05	5.70	6.20	6.90	17.15	31.30	52.70	86.60	98.85
100	5.35	5.10	5.60	5.85	5.35	3.85	7.50	27.80	50.65	81.90	99.25	100.00
200	5.15	5.25	5.75	4.65	5.15	4.75	10.75	48.50	78.80	98.30	100.00	100.00
						CCE	Pooled					
20	5.30	6.60	6.80	7.70	7.90	7.45	6.75	14.30	19.15	30.05	46.95	66.25
30	5.30	6.30	6.65	6.45	6.30	6.50	7.90	16.60	24.05	39.60	65.90	85.70
50	4.95	5.85	5.95	5.40	5.95	5.95	9.80	23.90	38.20	58.75	88.55	98.85
100	5.50	5.00	5.45	5.95	5.40	4.55	15.35	39.40	63.15	86.80	99.55	100.00
200	5.60	4.95	5.35	5.20	5.00	5.10	25.85	65.85	88.65	99.35	100.00	100.00
					F	Fixed Effe	cts estim	ator				
20	50.25	68.20	79.00	89.70	96.20	98.75	63.40	82.80	91.50	97.05	99.10	99.80
30	59.95	75.25	83.65	93.05	98.15	99.45	70.50	86.55	92.70	98.05	99.80	99.75
50	70.55	87.80	95.65	98.70	99.90	100.00	84.40	96.70	99.40	99.80	100.00	100.00
100	80.00	92.05	96.20	99.50	99.90	100.00	89.25	97.75	99.50	100.00	100.00	100.00
200	86.25	95.45	98.20	99.70	100.00	100.00	93.80	99.00	99.90	100.00	100.00	100.00

			$\frac{1}{\text{evel}}$ (×			1)	-		1	$\frac{100}{100}(H_1$).95)
			<u></u>	Г						Т		
						$\delta =$	- 0.8					
Ν	10	20	30	50	100	200	10	20	30	50	100	200
						CCE Me	ean Grou	up				
20	6.55	7.00	6.55	5.55	6.65	6.55	6.40	9.35	11.40	16.85	31.60	53.65
30	6.15	5.75	6.00	6.05	5.70	6.80	6.10	9.25	11.50	18.60	22.00	64.45
50	4.60	4.65	5.30	6.40	4.45	6.45	5.90	9.25	12.75	23.15	48.20	80.95
100	5.15	4.90	5.70	4.70	5.20	6.05	5.65	12.60	22.00	40.50	73.10	97.40
200	4.75	5.30	5.50	4.60	5.05	5.00	6.20	20.65	35.55	61.80	95.60	100.00
						CCE	Pooled					
20	4.60	6.70	7.05	6.35	7.40	7.15	5.55	9.15	11.95	17.65	33.85	54.45
30	5.20	5.80	6.75	5.70	5.30	7.30	6.65	10.05	13.05	20.30	26.55	66.15
50	5.15	5.50	5.15	6.85	5.05	6.10	6.85	11.15	15.70	26.90	52.00	81.70
100	5.45	5.05	5.30	5.65	4.95	5.55	8.45	15.95	26.55	45.40	76.85	97.60
200	5.30	5.60	5.20	4.75	4.75	4.90	11.55	29.00	44.25	70.80	96.60	100.00
					Fiz	xed Effe	cts estim	nator				
20	10.65	11.05	11.00	9.65	10.85	10.60	13.40	15.90	18.60	22.65	38.15	55.70
30	16.50	17.15	17.45	17.15	13.35	18.80	18.55	22.55	25.55	33.55	46.15	68.80
50	8.55	8.30	8.25	9.55	8.05	9.60	14.35	18.75	24.25	37.55	61.80	87.65
100	13.60	14.35	13.35	13.10	14.75	14.30	23.55	35.00	46.15	65.30	87.30	98.25
200	12.90	13.70	12.35	13.20	12.80	13.85	30.85	52.35	68.30	85.70	98.75	100.00

Table 2: Size and Power of CCE Type Estimators in Experiments B

	Siz	e (5% l	evel) (×	(100)(H	$\beta_1 = \beta_1 = \beta_1$	1)	Pov	wer (5%)	evel) (>	(100)(H)	$: \beta_1 = 0$.95)
			r -	Г						Т		
							$\delta = 0.4$					
Ν	10	20	30	50	100	200	10	20	30	50	100	200
						CCE	Mean G	roup				
20	5.55	7.35	6.50	6.35	5.75	6.90	6.75	12.00	15.90	25.90	51.80	80.45
30	5.80	6.05	5.90	5.45	6.90	6.50	7.15	13.50	19.40	34.60	64.30	93.40
50	4.80	4.50	4.55	6.15	5.50	5.90	6.15	14.80	25.15	47.75	84.50	99.00
100	4.65	5.00	5.45	5.20	5.65	5.25	7.75	26.25	44.80	77.10	98.45	100.00
200	5.15	5.60	5.80	4.65	5.60	4.75	9.75	44.15	73.65	96.85	100.00	100.00
						CC	E Poole	ed				
20	4.60	6.80	6.85	6.50	6.55	7.05	6.40	12.30	17.80	26.90	53.90	81.00
30	5.25	5.75	6.00	5.60	7.05	6.60	7.75	14.60	22.30	38.50	67.70	94.15
50	5.00	5.55	5.95	6.25	5.55	5.95	9.05	21.05	32.05	54.45	87.30	99.20
100	4.50	5.20	5.50	4.85	5.30	5.25	13.80	34.20	55.10	83.90	99.00	100.00
200	5.30	4.95	5.45	4.65	5.10	4.60	21.55	60.90	85.40	98.75	100.00	100.00
					F	Fixed E	ffects est	timator				
20	6.85	6.70	7.40	6.15	6.60	5.95	13.30	21.45	27.00	41.85	72.45	94.20
30	8.40	7.05	7.10	6.85	7.90	7.95	15.65	27.50	37.45	57.75	84.35	99.25
50	6.30	5.35	5.45	6.55	5.70	6.25	20.90	37.10	50.25	75.55	97.00	100.00
100	6.95	6.90	7.35	6.65	7.10	7.35	34.50	65.15	83.40	96.85	99.95	100.00
200	6.70	6.70	6.30	6.45	7.00	7.15	56.05	90.35	98.65	99.90	100.00	100.00

	5	Size (5%	level) (>	$\times 100)(H$	$I_0: \beta_1 = 1$	1)		wer (5%	level) (>	$\times 100)(H_1$	$\beta_1 = 0$.95)
				Т						Т		
						$\delta =$	0.8					
N	10	20	30	50	100	200		20	30	50	100	200
						CCE Me	ean Grou	ւթ				
20	5.95	6.30	7.80	7.85	6.30	6.65	6.85	9.30	13.25	17.40	30.20	46.90
30	5.70	5.60	6.05	6.50	5.75	6.65	6.50	8.90	12.25	19.35	35.00	62.50
50	5.60	5.75	5.25	5.80	5.30	5.80	6.65	9.25	15.30	24.25	49.05	79.05
100	5.55	5.20	5.25	5.35	4.60	4.95	5.45	12.25	22.00	37.50	7375	96.80
200	5.55	5.20	5.50	5.05	5.60	4.80	7.45	19.00	33.70	62.55	95.50	99.95
						CCE	Pooled					
20	4.85	6.85	7.05	7.50	6.60	7.05	5.95	10.60	14.10	19.55	31.40	48.40
30	5.35	6.35	5.85	6.80	6.15	6.80	6.60	10.95	13.85	20.90	36.95	63.00
50	5.80	4.95	5.60	5.45	5.55	5.75	7.40	12.10	17.80	27.20	52.40	80.90
100	4.65	5.20	5.55	5.55	4.60	4.80	8.00	17.00	27.40	43.30	77.90	97.85
200	5.50	5.25	4.15	4.50	5.65	5.35	12.75	27.70	44.25	70.55	96.00	99.95
					Fi	ixed Effec	ts estim	ator				
20	40.30	56.15	67.10	80.75	92.60	97.75	48.65	68.95	81.35	92.50	98.55	99.45
30	48.55	64.80	74.70	86.20	96.00	99.05	57.70	77.10	86.25	95.35	99.35	99.70
50	58.80	78.65	88.80	95.80	99.70	100.00	70.10	90.05	97.40	99.50	99.95	100.00
100	70.95	86.65	93.05	98.95	99.80	100.00	81.50	95.20	98.80	99.85	100.00	100.00
200	80.05	92.05	96.70	99.50	100.00	100.00	90.00	98.10	99.75	100.00	100.00	100.00

Table 3: Size and Power of CCE Type Estimators in Experiments C

	, Contraction of the second se	Size (5%	level) (\times	(100)(H)	$_{0}:\beta_{1}=1$.)	Po	wer (5%	evel) (>	$(100)(H_1)$	$: \beta_1 = 0.$	95)
			,	Г						Т		
						$\delta =$	0.4					
Ν	10	20	30	50	100	200	10	20	30	50	100	200
						CCE Me	an Grou	р				
20	5.90	6.85	7.15	8.40	7.05	7.35	7.05	12.05	17.15	25.80	44.45	65.30
30	5.00	5.95	7.20	6.75	6.00	6.75	6.85	12.55	19.15	33.20	59.80	85.55
50	5.90	5.75	5.05	5.20	5.80	6.10	7.35	15.40	27.80	47.90	83.40	98.25
100	5.50	5.20	5.30	5.95	4.90	4.25	6.85	24.25	45.25	76.25	98.75	100.00
200	5.65	5.45	5.85	4.70	5.60	5.05	10.35	42.45	71.55	96.70	100.00	100.00
						CCE I	Pooled					
20	5.05	6.85	7.10	7.80	7.25	7.10	7.05	13.75	19.10	28.20	45.15	63.80
30	5.70	6.80	6.60	6.65	5.85	7.00	7.65	15.90	22.95	37.25	62.30	83.90
50	5.75	5.40	6.00	5.45	5.65	5.95	10.00	21.05	33.85	54.60	85.80	98.35
100	5.15	5.05	5.45	5.90	5.40	4.30	14.45	35.70	56.95	82.65	99.00	100.00
200	5.90	5.05	5.30	4.45	5.60	5.05	23.40	58.50	83.85	98.35	100.00	100.00
					Fi	xed Effec	ts estima	ator				
20	49.25	67.45	78.15	89.55	96.05	98.65	61.50	81.25	91.05	96.90	99.25	99.75
30	57.70	74.05	82.70	92.60	97.90	99.45	69.45	86.60	92.45	98.00	99.80	99.75
50	69.25	87.15	95.40	98.50	99.90	100.00	82.15	95.95	99.30	99.85	100.00	100.00
100	79.15	91.60	96.10	99.45	99.90	100.00	88.70	97.60	99.45	100.00	100.00	100.00
200	85.80	95.40	98.00	99.70	100.00	100.00	93.80	98.95	99.90	100.00	100.00	100.00

		Size (5%	level) (>	$(100)(H_0)$	$\beta_1 = 1$)		wer (5%)	$\frac{1}{\text{level}}$ (×	$(100)(H_1)$	$: \beta_1 = 0$.95)
				Т	·					Т		
						Experim	ent D1					
Ν	10	20	30	50	100	200	10	20	30	50	100	200
					(CCE Mea	n Group)				
20	6.80	6.00	6.60	7.60	6.95	6.94	7.00	13.75	17.85	27.10	48.30	68.37
30	6.40	5.60	6.50	6.05	6.00	5.60	7.05	14.20	21.30	35.75	65.05	87.60
50	5.70	5.55	5.55	5.85	5.25	6.00	7.40	16.95	29.75	53.30	85.15	98.90
100	4.25	5.55	5.60	5.00	5.50	4.65	7.85	27.90	48.10	79.05	99.05	100.00
200	4.31	4.91	5.30	5.80	5.29	4.90	11.23	47.81	77.65	96.30	100.00	100.00
						CCE F	ooled					
20	6.30	6.80	7.00	7.25	7.70	7.76	8.00	15.05	19.75	28.80	50.35	69.08
30	5.65	5.90	6.40	6.65	6.00	5.85	6.65	18.05	24.20	40.80	67.25	88.40
50	4.15	6.00	5.70	5.60	4.80	6.40	10.30	22.80	36.20	58.80	87.80	0.98
100	5.10	6.05	4.95	5.60	5.55	4.55	15.05	37.95	59.60	85.60	99.50	100.00
200	5.57	4.70	4.70	5.75	4.97	4.72	23.48	64.37	87.95	98.15	100.00	100.00
					Fix	ed Effect	s estima	tor				
20	36.90	71.00	80.70	81.85	94.05	97.35	46.65	88.65	89.75	95.15	99.10	99.74
30	49.20	73.05	75.55	87.55	94.85	99.05	62.75	78.00	90.05	96.45	99.20	99.95
50	60.55	80.45	89.40	96.10	99.70	100.00	77.15	94.10	97.75	99.55	100.00	100.00
100	71.40	85.05	91.95	97.65	99.55	99.95	84.95	95.90	98.25	99.90	100.00	100.00
200	79.05	90.76	95.45	97.70	100.00	100.00	89.95	97.81	99.70	99.85	100.00	100.00

Table 4: Size and Power of CCE Type Estimators in Experiments D

		Size (5%	e^{1} level) (×	$(100)(H_0)$	$\beta_1 = 1$		Po	wer (5%	level) (>	$(100)(H_1)$	$: \beta_1 = 0.$	95)
			,	Г						Т		
						Experime	ent D2					
N	10	20	30	50	100	200	10	20	30	50	100	200
						CCE Mear	Group					
20	6.55	6.70	7.10	7.50	7.65	7.15	7.60	11.85	17.75	27.90	49.30	77.15
30	5.30	6.30	6.20	5.95	6.55	6.00	6.15	13.55	21.05	34.05	66.50	92.00
50	5.15	5.55	5.35	5.95	5.70	5.55	7.10	16.55	27.25	50.45	84.75	99.40
100	4.95	4.70	5.25	5.05	5.95	5.19	7.90	25.10	45.60	77.95	99.15	100.00
200	4.90	5.35	6.10	5.10	5.20	5.35	10.85	44.90	74.45	97.21	100.00	100.00
						CCE Po	ooled					
20	6.35	7.05	6.85	7.15	7.60	7.05	8.10	14.15	19.35	30.35	51.70	76.85
30	4.95	5.55	6.50	6.90	6.45	5.80	7.55	17.15	25.10	36.80	68.00	92.55
50	5.85	5.95	6.20	5.95	5.75	6.00	9.10	22.25	34.35	56.60	86.35	99.45
100	4.85	4.90	5.55	5.30	5.90	5.43	13.95	37.05	56.25	84.10	99.35	100.00
200	5.15	6.15	5.35	5.24	5.45	4.95	23.30	62.00	87.25	99.18	100.00	100.00
					Fiz	ced Effects	estimato	r				
20	29.20	35.10	49.00	57.30	78.90	91.40	44.45	58.25	74.40	86.55	97.25	99.55
30	40.20	45.50	57.10	66.15	85.35	94.10	52.20	67.60	80.15	89.40	98.80	99.90
50	47.30	54.75	68.90	81.65	93.70	98.90	66.55	80.35	92.15	98.35	99.90	100.00
100	59.85	66.00	76.00	86.20	96.30	99.64	78.40	88.75	95.70	98.90	99.95	100.00
200	68.60	73.55	85.05	92.93	98.80	100.00	87.25	94.40	97.80	99.73	100.00	100.00

			200		57.73	57.91	77.	.91	57.92		34.82	.85	13.48	72	3.33		20	71	71	1.39	69
			2(3 57.91													
			100		57.41	57.34	57.45	57.53	57.56		34.77	22.90	13.48	6.70	3.31		7.17	4.73	2.77	1.38	0.68
	100)	r .	50		56.81	56.36	56.65	56.82	56.68		34.67	22.71	13.50	6.69	3.31		7.20	4.69	2.79	1.38	0.68
	$\overline{\hat{r}}$ (×100)	T	30		55.77	55.75	55.68	55.64	55.91		34.43	22.85	13.28	6.66	3.28		7.18	4.79	2.73	1.38	0.68
nd C			20		54.55	54.31	54.32	54.68	54.99		34.12	22.49	13.43	6.61	3.27		7.09	4.71	2.82	1.38	0.68
A, B a			10		51.29	51.27	50.65	51.03	51.03		33.00	22.10	13.06	6.45	3.25		7.15	4.82	2.81	1.37	0.70
ents			0			87.95					-	01	08	.20	.41	-	34	69	68	38	97
perim			200		70.91	87.	114	162	231.04		61	76.01	99.	141	200.41		22.34	27.69	27.	51.	72.97
t in Exl			100	ors	49.80	61.41	80.40	114.41	162.08	$\delta = 0.8$	43.28	53.67	69.96	99.72	141.55	$\delta = 0.4$	15.67	19.54	25.44	36.27	51.50
and Average Correlation Coefficient in Experiments A, B and C	CD(p)	Τ	50	Experiment A: common factors	17.51	42.90	55.65	79.50	112.55	Experiment B: SAR errors with	30.55	37.85	49.34	70.31	99.82		11.11	13.74	17.91	25.50	36.25
tion Cc	CL	L ·	30	t A: com	13.12	32.24	41.89	59.63	85.15	SAR err	23.56	29.20	38.03	54.22	77.01	• •	8.57	10.61	13.77	19.60	27.87
Correla			20	perimen	10.27	25.50	33.43	47.47	67.00	ment B:	19.07	23.65	30.89	44.03	62.56	B:	6.88	8.53	11.18	15.89	22.57
rage			10	Ê	7.04	11.43	14.18		34.40	Experi	13.11				43.42	Experi	73	5.88	64	10.88	15.52
Ave			-	-							13	16.	21.	30.	43.		4	5.		10.	15.
			200		112.54	170.80	285.95	576.21	1155.45		67.87	67.40	66.73	66.90	66.46		14.03	13.88	13.88	13.78	13.67
Table 5: CD, $CD(p)$,			100		79.14	119.58	201.07	404.78	811.91		47.93	47.77	47.17	47.11	46.69		9.89	9.87	9.69	9.70	9.59
le 5: C			50		55.37	83.12	140.20	282.65	565.41		33.79	33.49	33.40	33.31	32.99		7.02	6.92	6.91	6.88	6.79
Tab	CD	T	30			63.69		•			25.99	26.10	25.45	25.65	25.37		5.42	5.47	5.23	5.31	5.22
			_														7	6	1	e S	6
			20		33.(50.6	85.(172.	346.92		21.(20.6	21.(20.79	20.6		4.3	4.3	4.4	4.33	4.2
			10		22.36	33.81	56.06	113.54	227.64		14.55	14.58	14.45	14.36	14.49		3.12	3.18	3.11	3.04	3.12
			Z		20	30	50	100	200		20	30	50	100	200		20	30	50	100	200

				<u> </u>										
		200		54.80	47.71	41.98	38.00	36.01		57.31	56.56	55.72	55.35	55.11
		100		54.47	47.30	41.74	37.68	35.66		56.97	56.01	55.40	54.97	54.73
100)	r .	50		54.17	46.70	41.20	37.11	34.97		56.42	55.08	54.63	54.25	53.87
\hat{r} (×100	T	30		53.57	46.09	40.42	36.00	34.28		55.45	54.46	53.68	53.05	53.08
		20		52.39	44.87	39.29	35.39	33.46		54.19	52.98	52.31	52.13	52.15
		10		50.67	43.04	36.87	32.30	30.20		51.20	50.13	49.15	48.16	48.16
		200	$\delta = 0.8$	66.10	82.01	106.86	152.44	216.39	$\delta = 0.4$	55.24	68.70	89.44	127.70	181.34
		100	Experiment C: common factors + SAR errors with $\delta = 0.8$	46.64	57.83	75.43	107.58	152.73	rors with	38.90	48.22	62.97	89.88	127.63
CD(p)	T	50	∟ SAR er	32.88	40.69	53.13	75.80	107.59	⊢ SAR er	27.30	33.68	44.10	62.98	89.25
CI		30	factors –	25.34	31.36	40.95	58.35	82.99	factors –	20.89	25.86	33.72	48.04	68.46
		20	common	20.45	25.37	33.20	47.36	67.34	common	16.73	20.70	27.07	38.73	55.18
		10	nent C: 6	14.14	17.48	22.81	32.70	46.47	nent C: 6	11.30	14.01	18.24	26.02	36.93
		200	Experir	106.83	140.71	207.80	378.08	7 718.34 46.47 67.34 82.99 107.59 152.73 216.39 3	Experir	111.72	166.83	275.81	550.72	1099.50
		100		75.09	98.65	146.07	265.09	503.07		78.53	116.81	193.90	386.73	772.10
CD	Ĺ	50		52.80	68.87	101.98	184.60	348.84		54.99	81.22	135.19	269.89	537.32
C	1	30		40.45	52.65	77.49	138.74	264.90		41.86	62.21	102.90	204.42	410.12
		20		32.30	41.85	61.50	111.36	211.06		33.41	49.42	81.88	164.03	328.99
		10		22.09	28.39	40.80	71.86	134.73		22.32	33.06	54.40	107.82	214.86
		Z		20	30	50	100	200		20	30	50	100	200

Notes: the CD, CD(p) and $\overline{\hat{r}}$ have been computed on residuals $\hat{u}_{it} = y_{it} - \hat{\alpha}_i d_{1t} - \hat{b}_P' x_{it}$, using the CCE Pooled estimator (42)

CESifo Working Paper Series

for full list see www.cesifo-group.org/wp (address: Poschingerstr. 5, 81679 Munich, Germany, office@cesifo.de)

- 2040 Rosario Crinò, Service Offshoring and White-Collar Employment, July 2007
- 2041 Carsten Hefeker and Michael Neugart, Labor Market Regulation and the Legal System, July 2007
- 2042 Bart Cockx and Muriel Dejemeppe, Is the Notification of Monitoring a Threat to the Unemployed? A Regression Discontinuity Approach, July 2007
- 2043 Alfons J. Weichenrieder, Profit Shifting in the EU: Evidence from Germany, July 2007
- 2044 Annika Alexius and Bertil Holmlund, Monetary Policy and Swedish Unemployment Fluctuations, July 2007
- 2045 Axel Dreher, Jan-Egbert Sturm and Jakob de Haan, Does High Inflation Cause Central Bankers to Lose their Job? Evidence Based on a New Data Set, July 2007
- 2046 Guglielmo Maria Caporale and Luis A. Gil-Alana, Long Run and Cyclical Dynamics in the US Stock Market, July 2007
- 2047 Alessandro Balestrino, It is a Theft but not a Crime, July 2007
- 2048 Daniel Becker and Michael Rauscher, Fiscal Competition in Space and Time: An Endogenous-Growth Approach, July 2007
- 2049 Yannis M. Ioannides, Henry G. Overman, Esteban Rossi-Hansberg and Kurt Schmidheiny, The Effect of Information and Communication Technologies on Urban Structure, July 2007
- 2050 Hans-Werner Sinn, Please Bring me the New York Times On the European Roots of Richard Abel Musgrave, July 2007
- 2051 Gunther Schnabl and Christian Danne, A Role Model for China? Exchange Rate Flexibility and Monetary Policy in Japan, July 2007
- 2052 Joseph Plasmans, Jorge Fornero and Tomasz Michalak, A Microfounded Sectoral Model for Open Economies, July 2007
- 2053 Vesa Kanniainen and Panu Poutvaara, Imperfect Transmission of Tacit Knowledge and other Barriers to Entrepreneurship, July 2007
- 2054 Marko Koethenbuerger, Federal Tax-Transfer Policy and Intergovernmental Pre-Commitment, July 2007
- 2055 Hendrik Jürges and Kerstin Schneider, What Can Go Wrong Will Go Wrong: Birthday Effects and Early Tracking in the German School System, July 2007

- 2056 Bahram Pesaran and M. Hashem Pesaran, Modelling Volatilities and Conditional Correlations in Futures Markets with a Multivariate t Distribution, July 2007
- 2057 Walter H. Fisher and Christian Keuschnigg, Pension Reform and Labor Market Incentives, July 2007
- 2058 Martin Altemeyer-Bartscher, Dirk T. G. Rübbelke and Eytan Sheshinski, Policies to Internalize Reciprocal International Spillovers, July 2007
- 2059 Kurt R. Brekke, Astrid L. Grasdal and Tor Helge Holmås, Regulation and Pricing of Pharmaceuticals: Reference Pricing or Price Cap Regulation?, July 2007
- 2060 Tigran Poghosyan and Jakob de Haan, Interest Rate Linkages in EMU Countries: A Rolling Threshold Vector Error-Correction Approach, July 2007
- 2061 Robert Dur and Klaas Staal, Local Public Good Provision, Municipal Consolidation, and National Transfers, July 2007
- 2062 Helge Berger and Anika Holler, What Determines Fiscal Policy? Evidence from German States, July 2007
- 2063 Ernesto Reuben and Arno Riedl, Public Goods Provision and Sanctioning in Privileged Groups, July 2007
- 2064 Jan Hanousek, Dana Hajkova and Randall K. Filer, A Rise by Any Other Name? Sensitivity of Growth Regressions to Data Source, July 2007
- 2065 Yin-Wong Cheung and Xing Wang Qian, Hoarding of International Reserves: Mrs Machlup's Wardrobe and the Joneses, July 2007
- 2066 Sheilagh Ogilvie, 'Whatever Is, Is Right'?, Economic Institutions in Pre-Industrial Europe (Tawney Lecture 2006), August 2007
- 2067 Floriana Cerniglia and Laura Pagani, The European Union and the Member States: Which Level of Government Should Do what? An Empirical Analysis of Europeans' Preferences, August 2007
- 2068 Alessandro Balestrino and Cinzia Ciardi, Social Norms, Cognitive Dissonance and the Timing of Marriage, August 2007
- 2069 Massimo Bordignon, Exit and Voice. Yardstick versus Fiscal Competition across Governments, August 2007
- 2070 Emily Blanchard and Gerald Willmann, Political Stasis or Protectionist Rut? Policy Mechanisms for Trade Reform in a Democracy, August 2007
- 2071 Maarten Bosker and Harry Garretsen, Trade Costs, Market Access and Economic Geography: Why the Empirical Specification of Trade Costs Matters, August 2007

- 2072 Marco Runkel and Guttorm Schjelderup, The Choice of Apportionment Factors under Formula Apportionment, August 2007
- 2073 Jay Pil Choi, Tying in Two-Sided Markets with Multi-Homing, August 2007
- 2074 Marcella Nicolini, Institutions and Offshoring Decision, August 2007
- 2075 Rainer Niemann, The Impact of Tax Uncertainty on Irreversible Investment, August 2007
- 2076 Nikitas Konstantinidis, Gradualism and Uncertainty in International Union Formation, August 2007
- 2077 Maria Bas and Ivan Ledezma, Market Access and the Evolution of within Plant Productivity in Chile, August 2007
- 2078 Friedrich Breyer and Stefan Hupfeld, On the Fairness of Early Retirement Provisions, August 2007
- 2079 Scott Alan Carson, Black and White Labor Market Outcomes in the 19th Century American South, August 2007
- 2080 Christian Bauer, Paul De Grauwe and Stefan Reitz, Exchange Rates Dynamics in a Target Zone A Heterogeneous Expectations Approach, August 2007
- 2081 Ana Rute Cardoso, Miguel Portela, Carla Sá and Fernando Alexandre, Demand for Higher Education Programs: The Impact of the Bologna Process, August 2007
- 2082 Christian Hopp and Axel Dreher, Do Differences in Institutional and Legal Environments Explain Cross-Country Variations in IPO Underpricing?, August 2007
- 2083 Hans-Werner Sinn, Pareto Optimality in the Extraction of Fossil Fuels and the Greenhouse Effect: A Note, August 2007
- 2084 Robert Fenge, Maximilian von Ehrlich and Matthias Wrede, Fiscal Competition, Convergence and Agglomeration, August 2007
- 2085 Volker Nitsch, Die Another Day: Duration in German Import Trade, August 2007
- 2086 Kam Ki Tang and Jie Zhang, Morbidity, Mortality, Health Expenditures and Annuitization, August 2007
- 2087 Hans-Werner Sinn, Public Policies against Global Warming, August 2007
- 2088 Arti Grover, International Outsourcing and the Supply Side Productivity Determinants, September 2007
- 2089 M. Alejandra Cattaneo and Stefan C. Wolter, Are the Elderly a Threat to Educational Expenditures?, September 2007

- 2090 Ted Bergstrom, Rod Garratt and Damien Sheehan-Connor, One Chance in a Million: Altruism and the Bone Marrow Registry, September 2007
- 2091 Geraldo Cerqueiro, Hans Degryse and Steven Ongena, Rules versus Discretion in Loan Rate Setting, September 2007
- 2092 Henrik Jacobsen Kleven, Claus Thustrup Kreiner and Emmanuel Saez, The Optimal Income Taxation of Couples as a Multi-Dimensional Screening Problem, September 2007
- 2093 Michael Rauber and Heinrich W. Ursprung, Life Cycle and Cohort Productivity in Economic Research: The Case of Germany, September 2007
- 2094 David B. Audretsch, Oliver Falck and Stephan Heblich, It's All in Marshall: The Impact of External Economies on Regional Dynamics, September 2007
- 2095 Michael Binder and Christian J. Offermanns, International Investment Positions and Exchange Rate Dynamics: A Dynamic Panel Analysis, September 2007
- 2096 Louis N. Christofides and Amy Chen Peng, Real Wage Chronologies, September 2007
- 2097 Martin Kolmar and Andreas Wagener, Tax Competition with Formula Apportionment: The Interaction between Tax Base and Sharing Mechanism, September 2007
- 2098 Daniela Treutlein, What actually Happens to EU Directives in the Member States? A Cross-Country Cross-Sector View on National Transposition Instruments, September 2007
- 2099 Emmanuel C. Mamatzakis, An Analysis of the Impact of Public Infrastructure on Productivity Performance of Mexican Industry, September 2007
- 2100 Gunther Schnabl and Andreas Hoffmann, Monetary Policy, Vagabonding Liquidity and Bursting Bubbles in New and Emerging Markets – An Overinvestment View, September 2007
- 2101 Panu Poutvaara, The Expansion of Higher Education and Time-Consistent Taxation, September 2007
- 2102 Marko Koethenbuerger and Ben Lockwood, Does Tax Competition Really Promote Growth?, September 2007
- 2103 M. Hashem Pesaran and Elisa Tosetti, Large Panels with Common Factors and Spatial Correlations, September 2007